# DRIGINAL

Declaration of Covenants, Conditions, Restrictions and Easements

Dated as of April 20, 1989

By

89208434

0000 M LaSalle National Bank As Trustee Under Trust Agreement Dated March 1, 1984 And Known As Trust No. 107701

Lasalle National Bank As Trustee Under Trust Agreement Dated September 1, 1988 And Known as Trust No. 213495 Orts Original Contractions of the Contraction of th



This instrument was prepared by and after recording should be. returned to:

Ira J. Swidler Katten Muchin & Zavis 525 West Monroe Street Suite 1600 Chicago, IL 60606

11 Car # 243199 DICU

Recorder - See last 2 pages for property where all tex mader listing.

# JA419190

Destatation of Covenancy, the Hillery, Partice of the contract of the contract

Dated as of April 20, 1998

Tagalle Rational Form Be Tructer Under Tract to Dated Stouth 1, 1 And Epoka &: Tract to 1970

Sec

. A Lebelle Mational cour As Truston Voler Trust More recu . A Date Asptester 1, 1946 And Arced as Trust Bo, 10205

e-version for beauties and property of the control of the control

Andrew Television (1994) The Constant Administration of the Constant Edit of Mandre Mandress, Substantial Half-Robert (1994)



my 6 mil 20 - 20 mil 20 30

1 SC # 50.361 DE

# INDEX TO DECLARATION OF COVENANTS, CONDITIONS, RESTRICTIONS AND EASEMENTS

| Article |  | Page  |
|---------|--|-------|
| I       | Definitions                                      | 2     |
| II      | Easements Benefitting the Residential Property   | 8     |
| III     | Easements Benefitting the Hotel Property         | 13    |
| IV      | Easements Benefitting the Commercial Property    | 17    |
| v       | Structural Support                               | 21    |
| VI      | Bullding Services                                | 23    |
| VII     | Compliance with Laws; Removal of Liens           | 27    |
| VIII    | Real Estate Taxes                                | 29    |
| IX      | Insurance  | 31    |
| X       | Maintenance and Repair; Damage to the Building   | 36    |
| XI      | Liens, Debts, Interest and Remedies              | 41    |
| IIX     | Arbitration                                      | 45    |
| XIII    | Unavoidable Delays                               | 47    |
| VIX     | Condemnation                                     | 48    |
| vv      | Limitation of Liability                          | 52    |
| XVI     | Architect  | 52    |
| XVII    | Depositary                                       | 54    |
| XVIII   | Disbursement of Funds by Depositary              | 56    |
| * XIX   | Estoppel Certificates                            | 59    |
| XX      | Association Acting for Unit Owners               | 60    |
| IXX     | Parking; Recreational Facilities; Master Antonna |       |
|         | System   | 61    |
| IIXX    | Alterations; Zoning                              | 64 7  |
| XXIII   | Notices  | 69. 4 |
| XXIV    | General  | 71 (  |
|         |  |       |

Mortgagee's Consent Exhibits and Appendices

# o de la comparta del comparta de la comparta del comparta de la comparta del la comparta de la comparta del la comparta de la

|  |  | efolda      |
|--|--|-------------|
| •  | north look                             |             |
|  |  |             |
|  | da prival Benefitzing th               | 71          |
|  | Basements Renefitting th               | IXX         |
|  | Faucharts Webstarting th               | VI          |
|  | i stoggus lagustus (                   |             |
|  | milding services                       |             |
|  | Compilance with terms; Ba              | ŢĬV         |
|  | Real Estate Taxes.                     | ITIV        |
|  | Transport Contract                     | XI          |
| agent to the art of manager of   | Maintenance and hepel                  | X           |
| · · · · · · · · · · · · · · · · · · ·  | Liens, Debts, Internati                | ΣX          |
| state of the state of the second   | Arbitestion                            | IIX.        |
| en e   | rot sidsblavenu                        | 1112        |
| Kananan kanan menangkan dianggan pengan  | Condepnation                           | VIX         |
|  | Lindta to constitution                 | <b>v</b> x. |
|  | · · · · · · · · · · · · · · · · · · ·  | 177         |
| ارد.<br>د م م و م د و م د و د و د و د و د و د و  | ······································ | 1177        |
|  | Debuffsenent of Lands by               | IIIVX       |
|  | Esteprel Certificates.                 | XIX         |
| in a second of the   | roi galios malislocaes                 |             |
|  | Parkings Regrestional F                |             |
| <ul> <li>A control of the contro</li></ul> |  |             |
|  | systemmetays                           |             |
| en e   | Alterations; Zoping                    | J.t.x.x     |
| and the state of t       | Not ices                               | TIIXX       |
|  | General                                | PXIV        |

Mortgagee's Consent Exhibits and Appendices

### DECLARATION OF COVENANTS, CONDITIONS RESTRICTIONS AND EASEMENTS

THIS DECLARATION is made and entered into as of the 20th day of April, 1989, by LaSalle National Bank, a national banking association, not personally but solely as Trustee under a Trust Agreement dated March 1, 1984, and known as Trust No. 107701 ("Trust 107701") and LaSalle National Bank, a national banking association, not personally, but solely as Trustee under a Trust Agreement clated September 1, 1988, and known as Trust No. 113495 ("Trust 1.3495") (Trust 107701 and Trust 113495 sometimes collectively called the "Declarant").

# RECITALS:

- A. The terms used in the Recitals, if not otherwise defined in the Recitals, shall have the meanings set forth in Article I hereof.
- B. Trust 113495 is the record legal title holder of the Total Parcel situated in Chicago, Cook County, Illinois and legally described in Exhibit 1.46 and leases the Total Parcel to Trust 107701 for a term of approximately seventy-five (75) years pursuant to a Lease Agreement dated october 7, 1988 (the "Ground Lease"), a Memorandum of which was recorded in the office of the Recorder of Deeds of Cook County, Illinois (the "Recorder") as Document No. 88464426. The Ground Lease provides that upon the expiration or sooner termination of the Ground Lease, title to the Commercial Building will automatically be in the Owner of the Commercial Parcel.
- C. Trust 107701 is the record legal title bolder of the Building.
- D. Immediately after this Declaration is recorded in the office of the Recorder, the Declarant will convey (i) the Residential Property to LaSalle National Bank, a national banking association, not personally but solely as Trustee under a Trust Agreement dated January 2, 1987, and known as Trust No. 111200 (the "Residential Property Trustee"), and (ii) the Hotel Property to LaSalle National Bank, a national banking association, not personally but solely as Trustee under a Trust Agreement dated January 2, 1987 and known as Trust No. 111150 (the "Hotel Property Trustee"). The foregoing conveyances (the "Conveyances") shall grant and reserve the Easements provided for herein and otherwise

#### DESCRIPTION OF THE PROPERTY OF RESTRICTIONS AND CASSASSANCE

The control of the state of the association, not personally but solely seemed the regard from the court of arrive beauty transport ("Truette 1077016) and lated in national back, a term describetton, processor (vitencorea don chaldeldos de Augusta de Augusta) (vitencorea don contratados de Augusta) (vitencorea do Augus Oppos thereto that the device departs that beingers coldectively sailed the Phosesser of

TAR TERME Deards used in the the Peritons, in the Recitars, shall have the re-

article of those for their Total Parcell withsend on 414 legally described in mulli Truck of the real forther depart pursuant to a Louis Coron vers Receptor of Coacto ែលជូនទេ១ ដែលម៉ាង Document, No. 38467 No. 12no A. Outli 128 . 1222 1 to 12 . explication or secretary taking and are not the first and are the first and the first or the fir Commercial Colorada and Sala antique of the Commercial Commercial Commercial Colorada Colorad Commercial Feloral

aregine egytes inosi, farener outh af 10770E fabru authitus.

alitative for the becomean the contact the contact of the Residential Property to [Lasally this see the deal of abaddiation, not perceptly ter selety on the contaction and the contaction of the co the mostdential property busineth, and the to (Takalla Kabipanil Bank, La auf Loui Loubenn and ma the end of the posterior as yields and yillshoused January, 2, 1987 and Jujown as Trusk and 1991 care they are the for the brong administrational and avison fits their

be subject to this Declaration. It is further intended that the Residential Property Trustee shall subject the Residential Property to the provisions of the Act by recording the Condominium Declaration with the Recorder.

- E. Each of the Commercial Building, the Hotel Building and the Residential Building may be structurally and/or functionally dependent on one or both of the others and may depend upon one or both of the others, to some extent, for structural support, enclosure, ingress and egress, utility services and certain other facilities and components necessary for the operation and use of the Commercial Building, the Hotel Building and the Residential Building.
- F. Decirrant desires by this Declaration to provide for the efficient operation of the Commercial Property, the Hotel Property and the Residential Property, to assure the harmonious relationship of the owners of each such Property, and to protect the respective values of each such Property, by providing for, declaring and creating certain easements, covenants and restrictions benefitting and burdening the Commercial Property, the Hotel Property and the Residential Property.

NOW, THEREFORE, the Declarant hereby declares that the Total Property and any part thereof is and shall be owned, held, mortgaged, transferred, assigned, sold, conveyed and accepted subject to this Declaration. Declarant does hereby further declare that this Declaration and each of the provisions, easements, covenants, conditions, rescrictions, burdens, uses, privileges and charges set forth herein or created hereunder shall exist at all times hereafter amongst, and be binding upon and inure, to the extent provided herein, to the benefit of, all parties having or acquiring any right, title or interest in or to any portion of, or interest or estate in, the rotal Property and each of the foregoing shall run with the land subjected to this Declaration.

#### ARTICLE I

#### <u>Definitions</u>

- 1.1 "Act" means the Condominium Property Act of the State of Illinois in effect on the date the Condominium Declaration is recorded, as amended from time to time.
- 1.2 "Architect" has the meaning set forth in Article XVI hereof.

the Residential Education as a second for the Residential Education as a second as a secon

efficient operation of the compensation of the owners of the content of the

Property and any pace there are a continued, and any pace there and property and any pace there are another a this declared, acted to this declared to this declared the continued that this declared and the acted the

#### I HIDTIMA.

#### encidention:

111 "Act" means the Condominium Property No. of the virial Tillingia in effect on the date the Condom of the little time to time.

1.2 "Architect" has the meaning set forth in Arrivia XVE

SCENE SON

- 1.3 "Association" means the 132 East Delaware Place Condominium Association, an Illinois not-for-profit corporation formed or to be formed for the purpose of administering the Condominium Property pursuant to the Act.
- 1.4 "Building" means all improvements, including, but not limited to, the footings, foundations, columns, piles, buildings, improvements, fixtures, equipment, machinery, Facilities, sidewalks, walkways, driveways and landscaping now or hereafter located in, on, under, within or upon the Total Parcel, including all alterations, rebuildings, replacements and additions thereto.
- 1.5 "Commercial Building" means the portion of the Building located wirin the Commercial Parcel.
- 1.6 "Commercial Easement Facilities" means Facilities located in the Residential Property or the Hotel Property (A) primarily benefitting the Commercial Property or the Owner of the Commercial Property, including, without limitation, those Facilities identified in Exhibit 1.6 as "Commercial Easement Facilities", or (B) necessary for the Owner of the Commercial Property to perform its chligations under Section 6.1 of this Declaration but in either case excluding Facilities, the Maintenance for which the Owner of the Residential Property is expressly responsible under Section 6.2 hereof or for which the Owner of the Hotel Property is expressly responsible under Section 6.3 hereof.
- 1.7 "Commercial Parcel" means that part of the Total Parcel legally described in Exhibit 1.7 attached hereto.
- 1.8 "Commercial Property" means the Commercial Parcel improved with the Commercial Building.
- 1.9 "Common Elements" means all portions of the Condominium Property except the Units.
- 1.10 "Condominium Declaration" means the Declaration of Condominium Ownership, Easements, Restrictions, Covenants and Bylaws for 132 East Delaware Condominium Association, as the same may be amended from time to time.
- 1.11 "Condominium Property" means the Residential Property from and after its submission to the Act and so long as it has not been withdrawn from the Act.
- 1.12 "Creditor Owner", except, where otherwise defined hereunder in a specific context, means an Owner to whom a payment of money or other duty or obligation is owed under this

- Condominium Association" means the the section of the Condominium Association; an Illinois the section of the borned formed for the section of the Condominium Evolution to be the the section of the condominium Evolution to the section of the condominium Evolution to the condominium Evolution to the condominium Evolution of the condominium Evolution to the condominium Evolution Evolution to the condominium Evolution Ev
- limited to, the fostings to the constitute of the constitute of the continue of the constitute of the
- located in the Massacrot Massacrot Marie Description the Massacrot Marie Description of the Commercial Property, the Commercial Property, the Massacrot Mass
- i die 11 des de 12 de 12
- Angroved with the commercial live on the control of the control of
- Property eraction of the montest reached for the me of the land that the Europeanty eraction that the traitment of the contest of the contest
- Candominian Demonstrate Decimal Reserving Condomination Demonstrates Reserving Reserving Condomination Condomination Description Description Condomination Description Descr
- From and after substantion by the between the last the last the between the first the substantion to the between the first the between which the between the betwe
- 1.12 Creditor Owner, canno per constant to the constant of the context of the con

本なないとはいい

Declaration by any other Owner who has failed to make such payment or to perform such duty or obligation as and when required hereunder.

- 1.13 "Declarant" has the meaning set forth in the preamble hereof.
- 1.14 "Declaration" means this Declaration of Covenants, Conditions, Restrictions and Easements dated as of April 20, 1989, including all exhibits, appendices, amendments and supplements thereto.
- 1.15 Defaulting Owner", except where otherwise defined hereunder in a specific context, means an Owner who has failed to make a payment of money owed under this Declaration to any other Owner or has inled to perform any of its duties or obligations as and when required under this Declaration.
- 1.16 "Depositary" has the meaning set forth in Article XVII hereof.
- 1.17 "Easements" means all easements provided for, declared or created pursuant to or in accordance with the terms and provisions of this Declaration.
- 1.18 "Emergency Situation" means (a) a situation impairing or imminently likely to impair structural support or Facilities of the Building; or (b) a situation causing or imminently likely to cause bodily injury to persons or substantial physical damage to all or any portion of the Building or any property within or about the Building; or (c) a situation causing or immediately likely to cause substantial economic loss to either the Owner of the Commercial Building or the Owner of the Hotel Property; or (d) a situation which materially interferes with the beneficial use of any Owner of its respective portion of the Building. The duration of an Emergency Situation shall be deemed to include the time reasonably necessary to remedy the Emergency Situation.
- 1.19 "Facilities" means all components, and any replacements or substitutions therefor, of the chilled and condense, water, central air handling and fan, temperature control, domestic water, fire suppression, sanitary waste, storm water, electrical, gas, detector and alarm, master satellite antenna, emergency power, telephone, elevator, escalator, lightning protection, kitchen waste and any other systems forming a part of the Building and designated or utilized to furnish utility or any other services to any portion of the Building, including without limitation: annunciators, antennae, boxes, brackets, cabinets, cables, coils, computers, conduits, controls, control centers, cooling towers, couplers, devices, ducts, elevator cars, equipment, fans,

Deblaration by any other Owner who had railed to make end or coloration perform such duty or obligation as say the presentation of the performance of the performance

Condition Passistations meters units forther, a constant of the conditions of the co

Lais "Detaulting Company, varous or hereundes in a specific construction of the constr

respectively the end that the menning was the first of the second of the

et suchtitutions therefor, of the call a conservation of the call of the call

# 69208434

# UNOFFICIAL COPY 4

fixtures, generators, hangers, heat traces, indicators, junctions, lines, machines, meters, motors, outlets, panels, pipes, pumps, radiators, risers, shafts, starters, switches, switchboards, tanks, transformers, valves, wiring and the like.

- 1.20 "First Commercial Mortgage" means the first mortgage or first trust deed in the nature of a mortgage, and all amendments, supplements and extensions thereto, on the Commercial Building (including, without limitation, any first mortgage or first trust deed which encumbers the Commercial Building and other property), together with any other mortgages or trust deeds (and all amendments, supplements and extensions thereto) on the Commercial Building held by the holder of the first mortgage or first trust deed on the Commercial Building.
- 1.21 "First Hotel Mortgage" means the first mortgage or first trust deed in the nature of a mortgage, and all amendments, supplements and extensions thereto, on the Hotel Property, together with any other mortgages or trust deeds (and all amendments, supplements and extensions thereto) on the Hotel Property held by the holder of the first mortgage or first trust deed on the Hotel Property.
- 1.22 "First Residential Montgage" means the first mortgage or first trust deed in the nature of a mortgage, and all amendments, supplements and extensions thereto, on the Residential Property or any part thereof made by the Declarant or the Residential Property Trustee. First Residential Mortgage does not include a first mortgage or first trust deed in the nature of a mortgage on a Unit made by a Unit Owner other than the Residential Property Trustee.
- 1.23 "First Mortgage" means the First Commercial Mortgage, the First Hotel Mortgage or the First Residential Mortgage, as the context requires. "First Mortgages" means the First Commercial Mortgage, the First Hotel Mortgage and the First Residential Mortgage.
- 1.24 "Garage" means the portion of the Garage Building designated or used for parking automobiles.
- 1.24A "Garage Building" means the portion of the Commercial Building bounded by North Rush Street, East Walton Street, East Delaware Place and North Ernst Court.
- 1.25 "Hotel Building" means the portion of the Building located within the Hotel Parcel.
- 1.26 "Hotel Easement Facilities" means Facilities located in the Commercial Property or Residential Property (A) primarily benefitting the Hotel Property or the Owner of the Hotel Property,

Fixtured, ganerators, hangers, host the distinction of the sale at the same of the sale at the sale at

Tiret truet deed in the netwer at a converse deed in truet truet truet truet deed in the netwer at a converse deed in the supplement of a converse deed and the supplement of a converse deed and the supplement of a converse deed and and a converse deed an

crust dead in the nature of the services of th

First timet doed in the main a common to the common of the supplements and extending the common of the supplements and extending the common of the supplement thereof made by the transfer of the supplement thereof made by the common took of the supplement of the common took of th

the First Hotel words as the start social will be social

design or used top parking out such as the safety of the design parking out such items

Pullding Bounded by Worth Runk Street, Fed we the service of Delaware Delaware Diace and Worth Street, Fed we the

logs whotel redicting mashis too postist of the control located within the Autob Enrost.

dies Hotel Easarent Pacifities" measured to the the Commarcial Property of Residential Property of the Commarcial Property of the

including, without limitation those Facilities identified in Exhibit 1.26 as "Hotel Easement Facilities", or (B) necessary for the Owner of the Hotel Property to perform its obligations under Section 6.3 of this Declaration, but in either case excluding Facilities, the Maintenance for which the Owner of the Residential Property is expressly responsible under Section 6.2 hereof or for which the Owner of the Commercial Property is expressly responsible under Section 6.1 hereof.

- 1.27 Intentionally Omitted.
- 1.25 "Hotel Parcel" means that part of the Total Parcel legally described in Exhibit 1.28 attached hereto.
- 1.29 "Hotel Property" means the Hotel Parcel improved with the Hotel Building.
- 1.30 "Hotel Property Trustee" has the meaning set forth in Recital D.
- 1.31 "Maintenance! means and includes operation, maintenance, repair, reconditioning, rejurbishing, reconfiguration, inspection, testing, cleaning, painting installation and replacement when necessary or desirable of Facilities (except repair and restoration required by Section 10.2 and 10.3 hereof) and includes the right of access to and the right to remove from the Building portions of such Facilities for any of the above purposes, subject, however, to any limitations set forth elsewhere in this Declaration.
- 1.32 "Owner of the Commercial Building' means the person or persons or entity or entities whose estates or interests, individually or collectively, aggregate, at any point in time, the present fee ownership of the Commercial Building. Owner of the Commercial Building does not include the owner of any remainder interest in the Commercial Building until such time as such interest becomes a present interest.
- 1.33 "Owner of the Commercial Parcel" means the person or persons or entity or entities whose estates or interests, individually or collectively, aggregate, at any point in time, fee simple ownership of the Commercial Parcel.
- 1.34 "Owner of the Commercial Property" means the Owner of the Commercial Building and the Owner of the Commercial Parcel.
- 1.35 "Owner of the Hotel Property" means the person or persons or entity or entities whose estates or interests, individually or collectively, aggregate, at any point in time, fee simple ownership of the Hotel Property.

including, without limitation shows would be real sources of the Exhtbit 1/26 as "Note: Research Perilises of the Hotel Property we sention 6.3 of this Daularuser, but the Pacifities, the Eniquenance for the Pacifities, the Eniquenance for the Property is expressly reasonable of the Valence of the Pacific The Country of the Country of

T. F. Intentionally Omitted.

1,18 Morel Parcell means that part of the Calledian Legally described in Exhibit 1.28 actached levers.

Tirrithotel Property second the fact server second to

redital December Property Transment has the Continue at the Property of the Property Transment of the Property of the Property

repair, reconditioning, referenced two contemporary reconditioning, referenced to the contemporary or desting the contemporary or destinated at a contemporary or destinated by Sautional contemporary of the contemporary contemp

persons of the Commercial Ball of the entity or entities there are all of the present individually or ralactively, equipment is a consecutive of the treasure of the consecutive of the treasure of the consecutive of the con

persons or entity or entities where etc. of the connectation when the consideration or entities where etc. of the conference of the connectation of the connectations.

the Commercial Englished Commercial Recognitive season of the control the Commercial Englishing and the Commercial Commercial Englishing Comm

Persons of the the third property seems that the persons of the third persons of the pe

- 1.36 "Owner of the Residential Property" means the person or persons or entity or entities whose estates or interests, individually or collectively, aggregate, at any point in time, fee simple ownership of the Residential Property. If and so long as the Residential Property has been submitted to and remains subject to the provisions of the Act, the Owner of the Residential Property shall mean all of the Unit Owners collectively and not individually.
- 1.37 "Owner" means the Owner of the Commercial Building, the Owner of the Hotel Property or the Owner of the Residential Property, as the context requires. "Owners" means the Owner of the Commercial Building, the Owner of the Hotel Property and the Owner of the Residential Property. Wherever the word "owner" appears herein and such word is not capitalized it shall mean any owner of any portion of the Total Property.
- 1.38 "Plans" means those architectural and engineering drawings and specifications used for the original construction of the Building which are identified in Exhibit 1.38 attached hereto. The Plans are not "as-built" plans and may be changed or supplemented as construction continues after the date this Declaration is recorded. Therefore, the Plans do not necessarily reflect the exact dimensions or location of each component in or about the Building.
- 1.39 "Recorder" has the meaning set forth in Recital B hereof.
- 1.40 "Residential Building" means the portion of Building located within the Residential Parcel.
- 1.41 "Residential Easement Facilities" means Facilities located in the Commercial Property or the rotal Property (A) primarily benefitting the Residential Property or the Owner of the Residential Property, including, without limitation, those Facilities identified in Exhibit 1.41 as "Residential Easement Facilities", or (B) necessary for the Owner of the Residential Property to perform its obligations under Section 6.2 of this Declaration, but in either case excluding Facilities, the Maintenance for which the Owner of the Commercial Property is expressly responsible under Section 6.1 hereof or for which the Owner of the Hotel Property is expressly responsible under Section 6.3 hereof.
  - 1.42 Intentionally Omitted
- 1.43 "Residential Parcel" means that part of the Total Parcel legally described in Exhibit 1.43 attached hereto.

- i.36 "Owner of the Residential Margorial Marity of Entity Entity Entity Entity Entity Entity Entity Entity Entity Shall seen all of the Entity Shall seen all of the Line Entity Shall seen all of the Line Entity Shally.
- Liff "Owner" means the concess of the contest of the Hotel tropers of Conservation Conservation Conservation Conservation Conservation Conservation Conservation of the Conservation and Such word as the Conservation of the the
- drawings and specifications the suitabline and specifications the suitable and specifications at suitable start that the supplemental as construction is recognized to the same time and the same time as a specification of the second start the same time and the short the same time.
- 1.39 Padada Market Mark
- 1.40 "Residentis" Colfant sease the colfant parties of the Las densial through
- Posted in the commercial Passess Principle Property In the Commercial Passess Principle Passes Passe
  - 1947 Intentionally caltean
- Tegality described in English to the party beat party of the color of

- 1.44 "Residential Property" means the Residential Parcel improved with the Residential Building.
- 1.45 "Residential Property Trustee" has the meaning set forth in Recital D.
- 1.46 "Total Parcel" means the parcel of real estate described in Exhibit 1.46 attached hereto (excluding the Building).
- 1.47 "Total Property" means the Total Parcel improved with the Building.
- 1.48 "Unit" means a part of the Condominium Property described as a "Unit" in the Condominium Declaration.
- 1.49 "Unic Owner" means the person or persons whose estates or interests, individually or collectively, aggregate fee simple ownership of a Unit Ownership.
- 1.50 "Unit Ownership" means a part of the Condominium Property consisting of one Unit and the undivided interest in the Common Elements attributable thereto.
- 1.51 "Valet Parking Area" means the portion of lower level 3 of the Building to be utilized for valet parking (identified as #LL301, #LL309 and #LL351 in the Flans).
- 1.52 Certain terms used in Exhibits 6.1(A) through 6.1(N) and 6.2(A) and 6.2(B) attached hereto are defined in Exhibit 1.52 attached hereto.

#### ARTICLE II

#### Easements Benefitting the Residential Property

- 2.1 The following perpetual easements burdening the Commercial Property and benefitting the Residential Property are hereby declared and created:
- (A) A non-exclusive easement in and to all structural members, footings, caissons, foundations, columns and beams and any other supporting components located within or constituting a part of the Commercial Property for the support of (i) the Residential Building and (ii) any Facilities located within the Commercial Property with respect to which the Owner of the Residential Property is granted an easement under this Declaration.

- 1.44 "Residential Property" rests that the state the same as a supported by the Residential college.
- 17 Rocaled D. Property Trades' has the meaning of the
- les First of the folial fermally means the govern or well a between the standard of the standa
- ar the Bullding of the Bullding and the the found the second of the seco
- described as a "Unit" means as poster of the contract of the described as a "Unit" in the conficultion described as
- or interacte, individually or collectively and collectively or collectively of the collectively.
- Property constants demanding and the contract of the contract
- This contain terms used in this to consider the consideration of the con

#### Bar approach

### Ling the constitution of the first third constitution of the const

- 221 far far fallowing parameters and a more than the far the f
- members, footings, caissons, toursell, and consect and contings, caissons, toursell, and consect and conting consect to the Commercial Property and till any with the Commercial Property and till any with the commercial Property and till any with the consect and the cons

- (B) A non-exclusive easement (i) for the use for their intended purposes of all Facilities at any time located in the Commercial Property and connected to Facilities at any time located in the Residential Property (and any replacements thereof) which provide or shall be necessary to provide the Residential Property with any utilities or other services or which may otherwise be necessary to the operation of the Residential Property, and (ii) permitting the exercise of the rights granted to the Owner of the Residential Property pursuant to Section 6.6(A) hereof during any period in which said rights may be exercised.
- (C) An exclusive easement to maintain encroachments in the event and to the extent that, by reason of the original construction of the Building, any reconstruction thereof, minor surveying errors, or the subsequent settlement or shifting of any part of the building, any part of the Residential Building encroaches or shall hereafter encroach upon any part of the Commercial Parcel. Such easement to maintain encroachments shall exist only as long as the encroaching portion of the Building continues to exist. In no event shall an easement for any encroachment be created in favor of the Residential Property if such encroachment unreasonably interferes with the reasonable use and enjoyment of the Commercial Property by the Owner of the Commercial Property.
- (D) An exclusive easement for the Maintenance of Residential Easement Facilities, subject to the rights of the Owner of the Commercial Building granted pursuant to Section 6.6(A) hereof.
- (E) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the truck ramps (identified as #160 and #LL283 in the Plans), the trucking concourse and loading dock (identified as #LL282 and #LL279 in the Plans) and the service corridors (identified as #LL271, #LL268, #LL265, #LL205, #LL215, #LL227 and #LL211 in the Plans) located in the Commercial Property, to and from the receiving room (identified as #LL289 in the Plans), the storage room (identified as #LL213 in the Plans) and the shafts housing elevators designated as Nos. 38, 39, 40 and 41 located in the Residential Property.
- (F) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through those sections of the common or fire stairways located, from time to time, in the Commercial Property.

- intended purposes of all Partities at any control of Commercial Property and connected partities at any control of the Residential Property and connected partition provides of shall be necessary with any untilities of the perty with any untilities of the perty and (ii) permitting the secrety and (ii) permitting the secrety and (ii) permitting the secret of the Araidense of the
- An excitation of the contractions of  $-(\mathfrak{O})_{\mathbb{N}^{-1}}$ the Jevent and to the oxcess that construction of the Bollding of menti silike oda ao teropani bahkenda part of the Buliding, end 4.600 on redinarion lists To Cashurators Commercial Parcelly Such opens into the occours and except to yino deaxe continues to exist. rtore on wir encine of the capables for a commission Arni Videnceserau daomiosorone doba streemed and to understand inc Commercial Property.
- Residential Basement Froilance, and the second for the second of Commercial Second Second of the Commercial Second of the Second of the Commercial Second of the Second
- pergons, material act equipment over a truck ramps (id. ritied as 100 and truck ramps (id. ritied as 100 and trucking donorarest and loading doct the flats) in the Plans, sind the Coasear at From the Coasear at From (identified as files and the room (identified as files and residential destinated as files in the Plans) and residential destinated as the Plans) at the Residential Brosorcy.
- persons, material and equipment over an end of the sections of the common of the state of the sections of the common of the tale, the sections of the commercial resperty.

TO A CONTRACT

- (G) A non-exclusive easement for the use of the trucking concourse and loading dock (identified as #LL282 and #LL279 in the Plans) located on lower level 2 of the Commercial Building for the receipt, transport, loading and unloading of materials and equipment and disposal of trash and for any other similar purposes for which such areas are customarily used in a residential or apartment building, subject, however, to the terms and conditions set forth in Exhibit 6.1(E) attached hereto.
- (H) A non-exclusive easement over, on, across and through the Commercial Property to the extent reasonably necessary (i) to permit the maintenance, repair, replacement, restoration or reconstruction of the Residential Property as required or permitted pursuant to this Declaration, or to the extent reasonably necessary to exercise the Easements set forth in this Section 2.1, (ii) during an Emergency Situation, or (iii) to construct and maintain substitute or additional structural support required by Article 7 hereof.
- (I) A non-exclusive easement for the use of the facade cleaning and repair platform and related equipment (the "Operator's Platform") located on the 67th level roof of the Building for the purpose of cleaning the exterior windows and facade of the Residential Building, or any portion thereof, and maintaining, repairing and replacing the exterior of the Residential Building (including facade, walls, windows, screens and the like), subject, however, to the terms and conditions set forth in Exhibit 6.1(J) attached hereto.
- (J) A non-exclusive easement for pedestrian ingress and egress over, on, across and through the Garage, the mall corridor (identified as #657 in the Plans) and vertibule (identified as #632 in the Plans) located in the Commercial Property to and from the elevator lobby (identified as #608 in the Plans) located in the Residential Property.
- (K) A non-exclusive easement for pedestrian ingress and egress over, on, across and through the vestibule (identified as #123 in the Plans) located in the ground level valet parking lobby in the Commercial Property to and from the vestibule (identified as #124 in the Plans) located in the ground level lobby in the Residential Property.
- (L) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the corridor (identified as #2914 in the Plans) and the stairs (identified as #19 in the Plans) located on level 29 in the Commercial Property to and from the mechanical rooms (identified as #2916 and #2919 in the Plans) located on level 29 in the Residential Property.

| ä.    | . : |                     | (, f | 11/1/17 |        | 20     | emikoe   | 11      | isuto     | Y 25-17 | 1257 . 15 | (6)      |         |         |
|-------|-----|---------------------|------|---------|--------|--------|--|---------|-----------|---------|-----------|----------|---------|---------|
| -at 0 | ÷   | $z_1 + \lambda z_2$ |      | \$ 150  |        | Zart Y | <b>对以股份</b>  | - pas k | back.     | DET 15  | 9771      | oonee    | ing     | truck   |
|       |     |                     |      | 1.      | 5      | 1      | $= \frac{1}{4} \left( \frac{1}{2} \left( \frac{1}{2} \left( \frac{1}{2} \right) \right) \right)$ | 710     | Haday     | 51 (    | Flans     | अ।इ      | n.t @   | 华山江之产   |
|       |     |                     |      | 3       |        |        | 12 0   | .7.12   | g and he  | dos     | : 6C.     | LOL      | · proi. | biiva   |
| 3.0   |     |                     |      | 100     | 1.0    |        | 1 12 31  | 31 M    | Section . | 3375    | en Lagra  | EMTS.    | eloi    | rodom   |
| - 1   | •   |                     |      |         |        |        | . • j. ** : *  |         |           | de v    | ស្រី និង  | នេសក្សាម | ব্ জ্ঞ  | limia   |
| .a :  |     |                     | ,    | 4.1     |        |        | . 11 452 - E   | A. (1)  | . • Om    | Seema.  | 2000      | no is    | it Ins  | bitann. |
|       |     | <del></del>         | ٠    | 100     | . 165. |        |  | Utilia  |           | กร่าง   | 4. 3.96   | anol     | ilbeo:  | o bas   |

- through the Commercial Property

  (i) to permit the mainterned, seconstruction of the Resident pursuant to this reasonably madesery to enact the Section 2.1, (ii) during an access to construct and maintain substates a consequence and maintain substates and maintain
- closming spd repair plants seed to seed to seed and repair plants of the seed repair plants of the seed of the see
- eqrags over, on, aurous and the construction of the quadratic of the construction as the construction the Plans Construction of the construction of the Residential Experts.
- egroes over, across and the very service of the ser
- persons; material and equipment over the leaves of the persons of the persons of the control of

- 2.2 The following perpetual easements burdening the Hotel Property and benefitting the Residential Property are hereby declared and created:
- (A) A non-exclusive easement in and to all structural members, columns and beams and any other supporting components located within or constituting a part of the Hotel Property for the support of (i) the Residential Building and (ii) any Facilities located within the Hotel Property with respect to which the Owner of the Residential Property is granted an easement under this Declaration.
- (B) A non-exclusive easement (i) for the use for their intended purposes of all Facilities at any time located in the Hotel Property and connected to Facilities at any time located in the Residential Property (and any replacements thereof) which provide or shall be necessary to provide the Residential Property with any utilities or other services or which may be otherwise necessary to the operation of the Residential Property and (ii) permitting the exercise of the rights granted to the Owner of the Residential Property pursuant to Section 6.6(A) hereof during any period in which said rights may be exercised.
- (C) An exclusive assement to maintain encroachments in the event and to the extent that, by reason of the original construction of the Building, any reconstruction thereof, minor surveying errors, or the subsequent settlement or shifting of any part of the Building, any part of the Residential Building encroaches or shall hereafter encroach upon any part of the Hotel Parcel. Such easement to maintain encroachments shall exist only as long as the encroaching portion of the Building continues to exist. In no event shall an easement for any encroachment be created in favor of the Residential Property is such encroachment unreasonably interferes with the reasonable use and enjoyment of the Hotel Property by the Owner of the Hotel Property.
- (D) An exclusive easement for the Naintenance of Residential Easement Facilities, subject to the rights of the Owner of the Hotel Property granted pursuant to Section 6.6(A) hereof.
- (E) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through those sections of the common or fire stairways located, from time to time, in the Hotel Property.
- (F) A non-exclusive easement over, on, across and through the Hotel Property to the extent reasonably necessary (i) to permit the maintenance, repair, replacement, restoration or reconstruction of the Residential Property as required or

| <b>#</b> 137(1) | olaskeni.     | 87585885840     | Landagade       | CHILWOLLESS! | nus Anadord    |
|-----------------|---------------|-----------------|-----------------|--------------|----------------|
| <b>\$</b> 600 € | <br>Strange M | - And Josephime | वर्ष वर्षा वर्ष | benefizin    | pun Aquadora   |
| ٠,              |               |                 |                 | realest:     | declared and c |

- members, columbna and heavy on the columbna is columbna. Columbna and heavy of the columbna and heavy of the columbna and heavy of the columbna and the columbn
- Antended purposes of all Tarifred to the first and purposes of all Tarifred to the property of the property of the provide the Rasidential Examples of the provide of the provide the provide of the provide the p
- Edge event and to the error construction of the error construction of the error construction of the entitle construction of the entitle construction of the entitle construction of the entitle construction entitle constr
- February (U) An exclusive earesens to the particle of the second of the
- peternia, incluidad and equapated authorization, etc., etc.,
- through the Hotel Property on the extent concerns to the the through the Hotel Property on the extent concerns the permit the halfstonesses, running remains the kellen the Regions that Trape

permitted pursuant to this Declaration, or to the extent reasonably necessary to exercise the Easements set forth in this Section 2.2, (ii) during an Emergency Situation or (iii) to construct and maintain substitute or additional structural support required by Article V hereof.

- (G) A non-exclusive easement for pedestrian ingress and egress over, on, across and through level 7 of the Hotel Property to and from the residential elevator lobby (identified as #708 in the Plans) located on level 7 of the Residential Property.
- 2.5 Each Easement created under this Article II which provides or requires, for its enjoyment, ingress and egress on, over, across or through the Commercial Property or the Hotel Property shall be subject (except in an Emergency Situation) to such reasonable limitations as the Owner of the Commercial Building or the Owner of the Hotel Property may, from time to time after consultation with the Owner of the Residential Property, impose with respect to the establishment of limited paths of ingress and egress and limited hours of the day or days of the week during which such Easements may be used to prevent any unreasonable interference with the use and operation of the Commercial Building and the Hotel Property and in order to assure the reasonable security of the Commercial Building and the Hotel Property; provided, however, that any such limitations shall not preclude or unreasonably restrict enjoyment or exercise of any such Easement.
- 2.4 Easements provided for, declared or created (a) under Section 2.1 shall be binding upon the Commercial Property and the Owner of the Commercial Property and (b) under Section 2.2 shall be binding upon the Hotel Property and the Owner of the Hotel Property, and, subject to the provisions of Article XX, all of such Easements (i) shall run in favor of and inure to the benefit of and be appurtenant to the Residential Property and (ii) if and so long as the Residential Property is submitted to the Act, shall be part of the Common Elements.
- 2.5 Each of the Owner of the Commercial Building and the Owner of the Hotel Property shall have the right, at its cole cost and expense, to relocate within its Property any Facilities and Easements which burden its Property and benefit the Residential Property, other than Easements declared or created under Sections 2.1(A), 2.1(C), 2.2(A) and 2.2(C), so long as such relocation does not have a material adverse effect on the Residential Property or the use of the Residential Property.

permitted pursuant to this neglection, or treatmently indeed and the section of t

(C) is non-avaluation of the vertical of the vertical of the constant of the c

provides or required for its solution of through its solution of through its solution of the s

Section 2.1 shall be riching the the the the the the burden of the Commercial be riching the the the the the birding upon to Heroper as (to be birding upon to Heroper to the continue of the property, and, suggested the saving the saving of such Essembles (t) shall the terminal to the saving to the beautiful to the common to the saving to the saving the form the form of the common to the saving to the common the barence to the common th

Pwhet Al the Rotel Proposity books need the Constitution of the Rotel Proposity books need the first proposition of the Chan Bacomenca declared and 2.1(A), 2.1(C), 2.2(A) and 2.1(C), so that have a material adverse established of the Residential Proposition of the Residential Proposition

#### ARTICLE III

#### Easements Benefitting the Hotel Property

- 3.1 The following perpetual easements burdening the Commercial Property and benefitting the Hotel Property are hereby declared and created:
- (A) A non-exclusive easement in and to all structural members, footings, caissons, foundations, columns and beams and any other supporting components located within or constituting a part of the Commercial Property for the support of (i) the Hotel Building and (ii) any Facilities located within the Commercial Property with respect to which the Owner of the Hotel Property is granted an essement under this Declaration.
- (B) A non-exclusive easement (i) for the use for their intended purposes of all Facilities at any time located in the Commercial Property and connected to Facilities at any time located in the Hotel Property (and any replacements thereof) which provide or shall be necessary to provide the Hotel Property with any utilities or other services or which may otherwise be necessary to the operation of the Hotel Property, and (ii) permitting the exercise of the rights granted to the Owner of the Hotel Property pursuant to Section 6.6(A) hereof during any period in which said rights may be exercised.
- (C) An exclusive easement to maintain encroachments in the event and to the extent that, by reason of the original construction of the Building, any reconstruction thereof, minor surveying errors, or the subsequent settlement or shifting of any part of the Building, any part of the Hotel Fuilding encroaches or shall hereafter encroach upon any part of the Commercial Parcel. Such easement to maintain encroachments shall exist only as long as the encroaching portion of the Building continues to exist. In no event shall an easement for any encroachment be created in favor of the Hotel Property if such encroachment preasonably interferes with the reasonable use and enjoyment of the Commercial Property by the Owner of the Commercial Property.
- (D) An exclusive easement for the Maintenance of Hotel Easement Facilities, subject to the rights of the Owner of the Commercial Building granted pursuant to Section 6.6(A) hereof.
- (E) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the truck ramps (identified as #160 and #LL283 in the Plans), the trucking concourse and loading dock (identified as #LL282 and #LL279 in the Plans), the service corridors (identified as #LL271, #LL268, #LL265 and #LL205 in the Plans) and the elevator vestibule

| ŧ** | 17  | 131   | 750 | 30 | 111  | 4   |
|-----|-----|-------|-----|----|------|-----|
|     | 1 ( | . 5 . | 2   | 2. | 7 27 | /·· |

|  | and the second s |  |
|--|--|--|
|  |  |  |
| _  |  |  |
| the account of the first   | and the state of t | the first of the second for the first of the second |
|  | Senefitting the  | Proc. 1 ( 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2  |
| المحادث والمرابط والمراجع والمراجع والمراجع والمراجع   | and the state of t | The way of a second contract of  |
| A compared to the second of the contract of th | and the second of the contract of the second | the control of the adjustment of the control of the control of   |
|  |  |  |

- er and the The Tollowing personnal common and anti-Commonstation Property and benefit the teath tracks the teath of the common and winder a common and co
- members, footings, collection searchent in and to all restrict a and members, footing, four elections and restrict and appointing components for a seal and all the commercial frequency for the leaguest of the commercial frequency for the commercial frequency
- intended numposed of all fact) its section is the following former all frequenced of all fact) its section is the following and conserved to the first local frequency (and see the first later) for any section of any substitution of the first local frequency to prove the first of the conserved for the first conserved for the first local frequency to the description of the first local frequency for the first local first local frequency for the first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local frequency for the first local first local frequency for the first local frequency for the first local first local frequency for the first local frequency frequency for the first local frequency for the first local frequency for the first local frequency frequency for the first local frequency frequency frequency for the first local frequency frequency frequency frequency frequency frequency frequency frequenc
- The event and to the event of a control of a control of the event of a control of the control of a control of the control of a control of the control of a contro
- [4] [4] [5] [6] An explusive examination the limit leaders of the land of the
- persons; material are reactually event to the event of th



(identified as #LL270 in the Plans) located in the Commercial Property, to and from the receiving room (identified as #LL294 in the Plans), the storage room (identified as #LL269 in the Plans) and the elevator shafts housing elevator Nos. 24, 25, 26, 27, 32 and 33 located in the Hotel Property.

- (F) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through those sections of the common or fire stairways located, from time to time, in the Commercial Property.
- (G) A non-exclusive easement for the use of the trucking concourse and loading dock (identified as #LL282 and #LL279 in the Plans) located on lower level 2 of the Commercial Building for the receipt, transport, loading and unloading of materials and equipment and disposal of trash and for any other similar purposes for which such areas are customarily used in a hotel building, subject, however, to the terms and conditions set forth in Exhibit 6.1(L) attached hereto.
- (H) A non-exclusive easement over, on, across and through the Commercial Property to the extent reasonably necessary (i) to permit the maintenance, repair, replacement, restoration or reconstruction of the Hotel Property as required or permitted pursuant to this Declaration, or to the extent reasonably necessary to exercise the Easements set forth in this Section 3.1, (ii) during an Emergency Situation, or (iii) to construct and maintain substitute or additional scructural support required by Article V hereof.
- (I) A non-exclusive easement for the use of the Operator's Platform located on the 67th level roof of the Building for the purpose of cleaning the exterior windows and facade of the Hotel Building, or any portion thereof, and maintaining, repairing and replacing the exterior of the Hotel Building (including facade, walls, windows, screens and the like), subject, however, to the terms and conditions set forth in Exhibit 6.1(3) attached hereto.
- (J) A non-exclusive easement for pedestrian incress and egress over, on, across and through the Garage and the mall corridor (identified as #657 in the Plans) located in the Commercial Property to and from the elevator lobby (identified as #631 in the Plans) located in the Hotel Property.
- (K) An exclusive easement for the use of the terrace portions of the level 30 roof of the Building.

(ddentified and flow the Plans) thest of the time of the trans of the transfer transfer the transfer transfer the transfer transfer the transfer transfer

perdonal material and equipment of the control of the control of the perdonal perdonal materials and equipment of the control of the control of the common of the control of the control of the common of the control of

through the commercial Property of the Commercial Property of the permit the commercial Property of the permit the construction of the parenty of the parenty of the parenty of the construction of the constr

A postatoria Piatiford Journel of the contract of the postatoria distributed on the contract of the postatoria postatoria contractoria postatoria contractoria postatoria contractoria cont

eqibesh over; con sakulusiyo osossat teg eesemi . It is eqibesh over; on the control of the cont

if it is the transfer commons the properties of the second portions of the less the sufference of the second of th

SON OF STREET

- (L) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the vestibule (identified as #123 in the Plans) located in the ground level valet parking lobby in the Commercial Property to and from the hotel lobby (identified as #126 in the Plans) located in the ground level lobby in the Hotel Property.
- (M) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the corridor (identified as #2914 in the Plans) located on level 29 in the Commercial Property to and from the mechanical room (identified as #2912 in the Plans) located on level 29 in the Hotel Property.
- (N) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the mall corridors (identified as #630 and #657 in the Plans), the vestibule (identified as #632 in the Plans) and the elevator lobby (identified as #621 in the Plans) located on level 6 in the Commercial Property to and from the elevator pits housing elevator Nos. 28, 29, 30, 31, 35, 36 and 37 (identified as #623, #615 and #628 in the Plans) located on level 6 in the Hotel Property.
- 3.2 The following perretual easements burdening the Residential Property and benefitting the Hotel Property are hereby declared and created:
- (A) A non-exclusive easement in and to all structural members, columns and beams and any other supporting components located within or constituting a part of the Residential Property for the support of (i) the Hotel Building and (ii) any Facilities located within the Residential Property with respect to which the Owner of the Hotel Property is granted as easement under this Declaration.
- (B) A non-exclusive easement (i) for the use for their intended purposes of all Facilities at any time located in the Residential Property and connected to Facilities at any time located in the Hotel Property (and any replacements thereof) which provide or shall be necessary to provide the Hotel Property with any utilities or other services or which may be otherwise necessary to the operation of the Hotel Property and (ii) permitting the exercise of the rights granted to the Owner of the Hotel Property pursuant to Section 6.6(A) hereof during any period in which said rights may be exercised.
- (C) An exclusive easement to maintain encroachments in the event and to the extent that, by reason of the original construction of the Building, any reconstruction thereof, minor surveying errors, or the subsequent settlement or shifting of any

- persons, saterial and captique concrete the former and the persons, saterial and captique of over, or correct assection of the light of the conservation looky (identified as first the light of the light of looky (identified as first the light of looky (identified as first the light) as the looky (identified the light) as the looky darked looky in the light light of the looky darked looky in the light of the light.
- paraons, instantal and equipment was easy to the temperature of paraons, instantal and equipment was the control of identified as first in the Commercial Property to the last tree. The Commercial Property to the last tree.
- persons material and continue to the continue of the continue
- ind 19.3. In The best failering over the large real large end of the large
- members; quibmes and inches established es
- intended parposes of all recification of a second distribution of all recification of a second distribution of the second distribution of the second distribution of the second distribution of the second of the second distribution of the second of the second distribution distributio
- the event of (C) the evaluative enactant to the second of the control of the subsequent enactance of the subsequent exclusion of the subsequent exclusion of the control of

part of the Building, any part of the Hotel Building encroaches or shall hereafter encroach upon any part of the Residential Parcel. Such easement to maintain encroachments shall exist only as long as the encroaching portion of the Building continues to exist. In no event shall an easement for any encroachment be created in favor of the Hotel Property if such encroachment unreasonably interferes with the reasonable use and enjoyment of the Residential Property by the Owner of the Residential Property.

- (D) An exclusive easement for the Maintenance of Hotel Easemert Facilities, subject to the rights of the Owner of the Residential Property granted pursuant to Section 6.6(A) hereof.
- persons, matitial and equipment over, on, across and through the portion of the Pasidential Property occupied by the shafts housing elevators designated as Nos. 38, 39, 40 and 41 as shown in the Plans and the adjacent elevator vestibules, including the right to use such elevators, located within the Residential Property on the terms and conditions sat forth in Exhibit 6.2(B) attached hereto.
- (F) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through those sections of the common or fire stairways located, from time to time, in the Residential Property.
- (G) A non-exclusive easement over, on, across and through the Residential Property to the extent reasonably necessary (i) to permit the maintenance, repair, replacement, restoration or reconstruction of the Motel Property as required or permitted pursuant to this Declaration, or to the extent reasonably necessary to exercise the Easements set forth in this Section 3.2, (ii) during an Emergency Situation or (iii) to construct and maintain substitute or additional structural support required by Article V hereof.
- (H) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and chrough the elevator lobby (identified as #4812 in the Plans) and corridor (identified as #4810 in the Plans) located on level 43 in the Residential Property to and from the elevator machine rooms (identified as #4809, #4815 and #4816 in the Plans) located on level 48 in the Hotel Property.
- 3.3 Each Easement created under this Article III which provides or requires, for its enjoyment, ingress and egress on, over, across or through the Commercial Property or the Residential Property shall be subject (except in an Emergency Situation) to such reasonable limitations as the Owner of the Commercial Building or the Owner of the Residential Property may, from time

part of the Bullding, any part of the House as the standing and shall become the standing and upon any part of the shall shall be maintain and the shall shall be maintain and the shall shall an easement for any considerable shall an easement for any considerable shall an easement for any considerable shall be shall b

- Particle (18 (19) An Antilusian entrance for the entrances of the second of the second of the Baachiful Action (18) and the second of the seco
- persons, material and requirement, or persons, nation of the Mestidential enquirement, or persons of the Mestidential enquirement of the Mestidential enquirement of the Adjacon alovation will be adjacon alovation of the Adjacon alovation of the Mestidential and conditions and conditions and conditions and conditions and conditions are the Mestidential and conditions and conditions are the Mestidential and conditions and conditions are the Mestidential and the Mestidential and
- through the Residential conservation of (2) who need the theory to the Residential conservation of (2) to pare to the conservation of reconstruction of the conservation of the conservati
- persons, moresial and equipment contact the policy of the property of the prop
- Bacon Basesont uracing under the constant of t



to time after consultation with the Owner of the Hotel Property, impose with respect to the establishment of limited paths of ingress and egress and limited hours of the day or days of the week during which such Easements may be used to prevent any unreasonable interference with the use and operation of the Commercial Building and the Residential Property and in order to assure the reasonable security of the Commercial Building and the Residential Property; provided, however, that any such limitations shall not preclude or unreasonably restrict enjoyment or exercise of any such Easement.

- 3.1 Easements provided for, declared or created (a) under Section 3.1 shall be binding upon the Commercial Property and the Owner of the Commercial Property and (b) under Section 3.2 shall be binding upon the Residential Property and the Owner of the Residential Property, and all of such Easements shall run in favor of and inure to the benefit of and be appurtenant to the Hotel Property.
- 3.5 Each of the Swner of the Commercial Building and the Owner of the Residential Property shall have the right, at its sole cost and expense, to relocate within its Property any Facilities and Easements which burden its Property and benefit the Hotel Property, other than Easements declared or created under Sections 3.1(A), 3.1(C), 3.2(A) and 3.2(C), so long as such relocation does not have a material adverse effect on the Hotel Property or the business operations conducted on the Hotel Property.

#### ARTICLE IV

### Easements Benefitting the Commercial Property

- 4.1 The following perpetual easements burdening the Residential Property and benefitting the Commercial Property are hereby declared and created:
- (A) A non-exclusive easement in and to all structural members, columns and beams, and any other supporting components located in or constituting a part of the Residential Property for the support of (i) the Commercial Building and (ii) any Facilities located in the Residential Property with respect to which the Owner of the Commercial Property is granted an easement under this Declaration.
- (B) A non-exclusive easement (i) for the use for their intended purposes of all Facilities at any time located in the Residential Property and connected to Facilities at any time located in the Commercial Property (and any replacements thereof)

impose; with respect to the creability of the second till second to the second in a second se

Condition Basements priviled to the control of the control of the Section Backlery B. Control of the Section Basement of the Character Basement and the control of the backlery backlery be about the control of the Backlery and old of the control of the control of the backlery backlery and old of the control of the backlery and old of the control of the backlery.

Eutopeatly:

2.5. Path of the Residential Parent of Communications of the Communication of the Residential Parent of Communication of the Residential Parent of Communication of the Communica

#### WI SHOTUMA

### Escuments included this the Composition requests

Remidential The Statiowass prospectual was ere leaded to a Remidential the State of the Color of

mentions tolumns and weams, and way rich auge a constant of tolumns and weams, and way rich auge a constitution of transling a part of the constitution of (1) the Commercial Pales, and a correct of the the Massachtial Inspector with earlier constitution of the Commercial Property is started as a constant that

tivistics (8) Acquire considerable considerable of the first order of the first of



. 5

which provide or shall be necessary to provide the Commercial Property with any utilities or other services or which may otherwise be necessary to the operation of the Commercial Property and (ii) permitting the exercise of rights granted to the Owner of the Commercial Building pursuant to Section 6.6(A) hereof during any period in which said rights may be exercised.

- (C) An exclusive easement to maintain encroachments in the event and to the extent that, by reason of the original construction of the Building, any reconstruction thereof, minor surveying errors, or the subsequent settlement or shifting of any part of the Building, any part of the Commercial Property encroaches or shall hereafter encroach upon any part of the Residential Parcel. Such easement to maintain encroachments shall exist only is long as the encroaching portion of the Building continues to exist. In no event shall an easement for any encroachment be created in favor of the Commercial Property if such encroachment increasonably interferes with the reasonable use and enjoyment of the Residential Property by the Owner of the Residential Property.
- (D) An exclusive easement for the Maintenance of Commercial Easement Facilities, subject to the rights of the Owner of the Residential Property granted pursuant to Section 6.6(A) hereof.
- (E) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the portion of the Residential Property occupied by the shafts housing elevators designated as Nos. 38, 39, 40 and 41 as shown in the Plans and the adjacent elevator vestibules, including the right to use such elevators, located within the Residential Property on the terms and conditions set forth in Exhibit 6.2(a) attached hereto.
- (F) A non-exclusive easement to employ the intermittent platform stabilization buttons, supports and accessories on the exterior surface of the Residential Building to tempsport the Operator's Platform to and from the several portions of the Building.
- (G) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through those sections of the common or fire stairways located, from time to time, in the Residential Property.
- (H) A non-exclusive easement over, on, across and through the Residential Property to the extent reasonably necessary (i) to permit the maintenance, repair, replacement, restoration or reconstruction of the Commercial Property as required or permitted pursuant to this Declaration, or to the

| ķι             | <br>: | , , -,- | phitoaq        | ខ្ទុំ មុខន        | 22950     | ង ១៨     | List       | <b>1</b> 集《文句》 | ebiyoz   | g do    | LOW    |
|----------------|-------|---------|----------------|-------------------|-----------|----------|------------|----------------|----------|---------|--------|
| ا بينه<br>اينه |       | 3.13    | neal wine      | ាង១៩៩ <b>១</b> ១។ | 30        | ittless  | Lita'      | Vine of        | it in    | rate ty | OIN    |
| 7.             |       |         | បានមេក ។ មេនជី | ap golfa          | utg skom  | CAST C   | og yg      | bakesbi        | ead o    | giyaê   | rit do |
| 7              |       |         |                |                   | Sea iso   | i oktoke | 1 0113     | mon in the i   | गामक्य । | (11)    | bris   |
| •              |       |         | ing a second   | ga szálaz         | 7 3 4 13  | មខារស្ន  | <b>一种推</b> | I HOLL         | stones   | Const   | the    |
| 4. 1           |       |         |                |                   | gigger da |          | D. Kois    | activi         | ni bo    | lasg.   | YME    |
|                |       |         |                |                   | -         |          |            |                |          |         |        |

- the event and to the exert test.

  construction of the Building and as a serie of the Building and as a serie of the Building as a serie of the Building as a serie of shall be as a serie of analy as the series of anily as the series of as the continues to exist.

  continues to exist. In series of another as the series of another as a series of another as a series as a series of another as a series as a seri
- Establication (D) An exclusive consideration of the Commencial Establication of the Commencial
- persons, aaterial and error to the independence of the modern and error and error of the Resident of the Resident of the Resident of the Resident of the Standard of the address of the Resident of the Standard of the Resident of the Reside
- platicing (I) A conversion of consequent of the platicing of the property of the consequent of the consequence o
- persons, material and equipment ever, or or or or the sections of the common or fire scattery of the common or fire scattery.
- through the Sasidenaits Droperty of the party of the sasidenaits of the party (i) to party the mannerative of the party (i) to party the mannerative of the restoration of the required of party that the construction of the cons

extent reasonably necessary to exercise the Easements set forth in this Section 4.1, (ii) during an Emergency Situation, or (iii) to construct and maintain substitute or additional structural support required by Article V hereof.

- (I) A non-exclusive easement for the use of risers and shafts within the Residential Property for the extension of Facilities necessary for the operation of a satellite receiving station, antennas and any other related equipment on the 67th level roof of the Building.
- 4.2 The following perpetual easements burdening the Hotel Property and benefitting the Commercial Property are hereby declared and created:
- (A) A non-exclusive easement in and to all structural members, columns and beams, and any other supporting components located in or constituting a part of the Hotel Property for the support of (i) the commercial Building and (ii) any Facilities located in the Hotel Property with respect to which the Owner of the Commercial Property is granted an easement under this Declaration.
- (B) A non-exclusive comment (i) for the use for their intended purposes of all Facilities located in the Hotel Property and connected to Facilities located in the Commercial Property (and any replacements thereof) which provide or shall be necessary to provide the Commercial Property with any utilities or other services or which may otherwise be necessary to the operation of the Commercial Property and (ii) permitting the exercise of rights granted to the Owner of the Commercial Building pursuant to Section 6.6(A) hereof during any period in which said rights may be exercised.
- (C) An exclusive easement to maintain encroachments in the event and to the extent that, by reason of the original construction of the Building, any reconstruction thereof, minor surveying errors, or the subsequent settlement or shifting of any part of the Building, any part of the Commercial Property encroaches or shall hereafter encroach upon any part of the Hotel Parcel. Such easement to maintain encroachments shall exist only as long as the encroaching portion of the Building continues to exist. In no event shall an easement for any encroachment be created in favor of the Commercial Property if such encroachment unreasonably interferes with the reasonable use and enjoyment of the Hotel Property by the Owner of the Hotel Property.
- (D) An exclusive easement for the Maintenance of Commercial Easement Facilities, subject to the rights of the Owner of the Hotel Property granted pursuant to Section 6.6(A) hereof.

- extent reasonably nacessary to exercise the Ensemble set that Section 4.1. (11) during an Keargency Michaels of the Construct and maintain substitute of a district of Article V hereof.
- shaits within the Pasidential Property (of the Control of Facilities ascessory for the operation of American and Station, and Station, antennas and any other related new reserved the Sulising.
- 4.2 The iollowing perpetual easement the case the constitution of the constitution of
- (A) A con-exclusive ascent to the members, columns and beams, and asset of the constituting a part of the constituting a part of the constitution will include a lace of the Hotel Property will asset to the Commercial Property will asset to the Commercial Property is quarter the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property is quarter to the Commercial Property in the Commercial Property
- intended purposed of All Farity of the the the second of and and and another the translated for the the the second of the the description of the the the the description of the provider the description of the translated for the description of the description of
- the event and to the ercent that he can be not to the construct to the ercent that he can be not to the construct the construct the construct and the construct to the construct to the construct to the construct and the construct of the construct and the constructions of the construct and the constructions of the construction of the construction
- Commercial Baremant Facilibes, exclusive basement to the the service of the service of the commercial baremant Facilibes, exploses the testing of the content of the conten

- (E) A non-exclusive easement to employ the intermittent platform stabilization buttons, supports and accessories on the exterior surface of the Hotel Building to transport the Operator's Platform to and from the several portions of the Building.
- (F) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through those sections of the common or fire stairways located, from time to time, in the Hotel Property.
- (G) A non-exclusive easement over, on, across and through the Hotel Property to the extent reasonably necessary (i) to permit the maintenance, repair, replacement, restoration or reconstruction of the Commercial Property as required or permitted pursuant to this Declaration, or to the extent reasonably necessary to exercise the Easements set forth in this Section 4.2, (ii) during an Emergency Situation, or (iii) to construct and maintain substitute or additional structural support required by Article V hereof.
- (H) A non-exclusive easement for ingress and egress by persons, material and equipment over, on, across and through the corridors (identified as #2130 and #3138 in the Plans) and elevator lobby (identified as #3137 in the Plans) located on level 31 of the Hotel Property to and from the elevator machine room (identified as #3146 in the Plans) located on level 31 of the Commercial Property.
- 4.3 Each Easement created under this Article IV which provides or requires, for its enjoyment, ingress and egress on, over, across or through the Residential Property or the Hotel Property, shall be subject (except in an Emergency Situation) to such reasonable limitations as the Owner of the Residential Property or the Owner of the Hotel Property may, from time to time after consultation with the Owner of the Commercial Building, impose with respect to the establishment of limited paths of ingress and egress and limited hours of the day or days of the week during which such Easements may be used to prevent any unreasonable interference with the use and operation of the Residential Property and the Hotel Property and in order to assure the reasonable security of the Residential Property and the Hotel Property; provided, however, that any such limitations shall not preclude or unreasonably restrict enjoyment or exercise of any such Easement.
- 4.4 Easements provided for, declared or created (a) under Section 4.1 shall be binding upon the Residential Property and the Owner of the Residential Property and (b) under Section 4.2 shall be binding upon the Hotel Property and the Owner of the Hotel

platform etechlization buttons, supports to copie the inches platform etechlization buttons, supports the second control of the seco

persons, material and oquipment over the person of the common or live other or the test of the Hotel Property.

through the Hotel Property to the eneme of the England the Hoperty to the eneme of the Leconstruction of the Conservation of the Conservation of the Conservation of the Satisfaction of the Satisfaction of the Hardison of H

persons renterial and entirement of the control of

provides or requires to the release to the provides or requires to the release or county, earlies to the release of the relation of the release of the relation of the relat

Section 4.4 Edgenors provided top, declered to recover the section 4.1 Statistics provided top, declered to the section of the Section Property and the section that the Hotel Property and the control of the Declered the Section that Hotel Property and the Ounge of the Section Charles are sections.

Property, and all of such Easements shall run in favor of and inure to the benefit of and be appurtenant to the Commercial Property.

4.5 Each of the Owner of the Hotel Property and the Owner of the Residential Property shall have the right, at its sole cost and expense, to relocate within its Property any Facilities and Easements which burden its Property and benefit the Commercial Property, other than Easements declared or created under Sections 4.1(A), 4.1(C), 4.2(A) and 4.2(C), so long as such relocation does not have a material adverse effect on the Commercial Property or the business operations conducted on the Commercial Property.

#### ARTICLE V

### Structural Support

- 5.1 No Owner for the Owner of the Commercial Parcel shall take any action which would adversely affect the structural safety or integrity of the Building.
- 5.2 If for any reason the structural support for any portion of the Building is hereafter reduced below the support required to maintain the structural safety or integrity of the Building, the Architect shall review, at the request of any of the Owners or the Owner of the Commercial Parcel, the extent of any such reduction and the need for or adequacy of any such substitute or additional structural support. The Architect shall also estimate, if possible, the time reasonably necessary to provide adequate substitute or additional structural support.
- 5.3 If substitute or additional structural support is required in a portion of the Building in which the structural support shall have been reduced, then the Owner or Owners responsible for such reduction, if the responsible Compare or Owners can be determined, shall commence the construction of such substitute or additional support within a reasonable time under the circumstances, and having commenced such construction shall proceed diligently to cause the completion of such construction in accordance with plans and specifications prepared by or approved by the Architect and approved by the Owners, the Owner of the Commercial Parcel and the holders of the First Mortgages. The responsible Owner or Owners shall pay all costs and expenses, including, without limitation, the Architect's and any other architectural fees, in connection with construction of the substitute or additional support. The Owners shall attempt in good faith to determine which Owner or Owners are responsible for such reduction, which determination shall be subject to the approval of (i) the holder of the First Commercial Mortgage in the

Property, Sand (all of such Samenante coal) each in the invire to the policite of and bar exercise and other Property.

to report this after a new country to this to him the country of t the Residential Property shall have "Lee the all cotto etapolar of anabora birs Easements and oh burden its lunger to Froperty, Other than Easonous a double of a. ri(ar, laving), a. r(a) and acress, he required not have a material adverse countries 1000 M the business operations contoured on the law ere

#### 11.70117条点

Burelle Oundaminate Company of the Company take any metaton which would as reverly or integrity of the fullding.

are the roses any reason the error of the Building is hereafthe and and maintain the structural safely o Architect shall revive, at the Owner, of the Commercial Pole and the pead for at adequed structural support. TO STALL SAN ismum sada laasidibha aa arudidaduk

of the transport of the contract of the contra required in a portion of the mall support shall bave been present to the receptor of the recep Control of the Water States can be distormined. incomentation of the states and upper dental sinks and sparational sections. ware there we are integrate and essent the transpirit to appoin accordance which plans and squeet and see the 117 by the Architoot and approved by the sections 44 Commercial Paynel and the holders of the conresponsible Concentration are supposed to the contraction of including, without limitation, the colorest water and appropriate the colors of the co architectural feer, in substitute for additional suppost. The first of the good failth to delegation which Geres or become the tree of and such realistion, which idekersing the little see the see approvate of (4) chainelder of the Farse Conserved Livery or to

TO TO THE STATE OF THE STATE OF

7

event that the Owner of the Commercial Building is determined to be responsible either alone or together with any other Owner, (ii) the holder of the First Hotel Mortgage in the event that the Owner of the Hotel Property is determined to be responsible either alone or together with any other Owner, and/or (ii) the holder of the First Residential Mortgage in the event that the Owner of the Residential Property is determined to be responsible either alone or together with any other Owner. If such parties are unable, within thirty (30) days after such reduction is discovered, to agree which Owner or Owners are responsible for such reduction, the Owners shall request the advice of the Architect. If after receiving the Architect's advice, the Owners cannot agree which Owner or Owners are responsible for such reduction, then such determination shall be made by arbitration pursuant to Article XII hereof.

- 5.4 In the event that the Owner or Owners determined responsible for the reduction in structural support fail to commence the construction of substitute or additional support within a reasonable time under the circumstances, or having commenced such construction fails to proceed diligently to its completion, any Creditor Owner shall have the right to complete the construction of such substitute or additional support at the expense of the Defaulting Owner, and all costs and expenses incurred by Creditor Owner shall be due from the Defaulting Owner on demand.
- 5.5 If delay in constructing substitute or additional support would endanger the structura? safety or integrity of the Building, then, without regard to which Comer or Owners shall be determined responsible for the reduction, the Owner of the portion of the Building in which the reduction shall have occurred or is then occurring shall, upon not less than ten (10) days advance written notice to the other Owners (except that such advance written notice shall not be required in an Emergency Situation), provide substitute or additional structural support wherever may be required, or the Owners shall together undertake to provide substitute or additional structural support; provided, however, the responsible Owner shall be liable for and pay all costs and expenses incurred as a result of any required substitute or additional support. If the responsible Owner or Owners cannot be determined, or if all Owners are responsible, or if the reduction in structural support results from a defect in the original construction or design of the Building, an act of God or force majeure, then the Owners shall share the cost of providing substitute or additional structural support, including, without limitation, any fees of the Architect, in the manner agreed to by the Owners. If the Owners cannot agree, the Owners shall request the advice of the Architect. If after receiving the Architect's advice, the Owners cannot agree on the sharing of such costs, then

event that the Owner of the Comercial and the responsible either alone or comment the holder of the First water or the holder of the First water or together vith any occasion together with any other within thirty (30) days of some agree which Owner or Comercial request is request to Comercial request in the Owners shall request in the Owners shall request in the Owners are responsible to the Archivect in Owners are responsible to the Owners are responsibl

respensible for the constant of the commence the constant of the constant of the constant of the constant of the commence such constant of the commence such constant of the committee constant of the constan

Supplied dolays of the continues of one make english and an and any any and an anganta bidan anaque and the state of t determined ramponrable for the true the of the public and dealer all painting odd to taen iquoseriang is lake junta suotojanant written softice to the cities course written motice shall not be respiced provide respectivity or radio energy Wherever say he required to the overly the configuration to provide substitute as admiitudus afarosa ot nowever the temperations of the converse gosts and expenses inchrimed a province of any 1 organistic entropy and it branquis is no builded De determined, or if all our cas are are geduction in Structural sampert read to original construction or section of the force (majeune, then the Owners and the substitute or additional amount of an an limitetion, and coes of the Armitet andimil the Owners. It instrumers wormant mades, where the the advice of the Architect. If sizer recovered adviced pine of the contract agree on the contract of the

89200434

such determination shall be made by arbitration pursuant to Article XII hereof. The foregoing shall not be deemed to limit any rights any of the Owners may have against third parties.

#### ARTICLE VI

#### Building Services

- 6.1 The Owner of the Commercial Building shall furnish or cause to be furnished the following services to the Owner of the Residential Property and the Owner of the Hotel Property when, as and if required:
- (A) Chilled Water. Production and circulation of chilled water upon the terms and conditions set forth in Exhibit 6.1(A).
- (B) Emergency Power. Emergency electrical service upon the terms and conditions set forth in Exhibit 6.1(B).
- (C) City Water. Supply city water upon the terms and conditions set forth in Exhibit 6.1(C).
- (D) Fire Alarm and Emergency Telephone Systems. Maintenance of Building's facilities which monitor the Building's fire alarm and emergency telephone systems upon the terms and conditions set forth in Exhibit 6.1(D).
- (E) Loading Dock and Trash Removal. Maintenance of (i) the service area and loading dock and all roadways, ramps and other means of access thereto located within the Commercial Property and (ii) trash removal service for compacted trash, all upon the terms and conditions set forth in Excipit 6.1(E).
- (F) Exterior Lighting. Maintenance of all exterior lighting of the Building upon the terms and conditions set forth in Exhibit 6.1(F).
- (G) Facade. Maintenance of the Building facade upon the terms and conditions, and solely to the extent, set forth in Exhibit 6.1(G).
- (H) Security. Provide continuous security service for the loading dock and service entrance and monitor certain portions of the Building and all stairwell alarms upon the terms and conditions set forth in Exhibit 6.1(H).
- (I) Street Level Exterior Maintenance and Snow Removal. Snow removal and Maintenance of the canopy over the street level entrance to the Hotel Lobby and Residential Lobby, and curb cuts,

| such determination shall be made by there he was the torest and whalf are the forest not the forest nave and the forest way way and the forest way |
|--|
|  |
|  |
| dause to be fureished the following control of the file following control of the following control of the file flat of the following control of the file function of the file  |
| (A) Chilled Water Thomas Ind Continued Chilled Water Upon the Lind Cond Cont. S. C.  |
| (B) Endroency Power. Entropy Co. (B) the terms and conditions set to be in the co. (B).  |
| (C) City Water. Sandly of Commission of Conditions Sat forth in twicks a. 17   |
| Maintenance of Building's Lealing's Lealing Sections and Exhibit Lealing L |
| (a) Loading Dack are con to be   |
| (F)  |
| (G) Fracede. Maintennaço es es esta de la facede de la fa |
| the loading dock and service are read a read to the control of the |

10.00 BC

entrance to the Hotel Laber and Lagidanced cotty on the cotty

Snow remark lavel havel have thing must be the second

conditions sat forth in Exhibit

sidewalks, trees, tree lights, tree pits, stone containers and other improvements along North Michigan Avenue, North Rush Street, East Walton Street and East Delaware Street and Maintenance of Ernst Court between East North Walton Street and East Delaware Street, upon the terms and conditions set forth in Exhibit 6.1(I).

- (J) Operator's Platforms. Maintenance of the Operator's Platforms upon the terms and conditions set forth in Exhibit 6.1(J).
- (K) Roofs. Maintenance of the various roofs of the Building, upon the terms and conditions set forth in Exhibit 6.1(K).
  - (L) Intentionally Omitted.
- (M) Fire Suppression System. Maintenance of the fire suppression system for the Hotel Property and the Residential Property upon the terms and conditions set forth in Exhibit 6.1(M).
- (N) Secondary Condenser Water System. Maintenance of the secondary condenser water system for the Hotel Property and the Residential Property upon the terms and conditions set forth in Exhibit 6.1(N).
- 6.2 The Owner of the Residential Property shall furnish or cause to be furnished the following pervices to the Owner of the Commercial Building and the Owner of the Hotel Property when, as and if required:
- (A) Water Redelivery. Redelivery of water upon the terms and conditions set forth in Exhibit 6.2(1).
- (B) Elevator Service. Maintenance of service elevator No. 41 (as shown on the Plans) and passenger elevators Nos. 38, 39 and 40 (as shown on the Plans) when service elevator No. 41 is unavailable, all upon the terms and conditions set forth in Exhibit 6.2(B).
  - 6.3 Intentionally Omitted.
- 6.4 Each Owner shall make a good-faith effort to operate its Facilities and furnish all services as required under this Article VI in a manner which will provide each Owner with comfortable occupancy and enjoyment of its respective portion of the Building for its intended use as first-class commercial, hotel or residential property, but in no event shall an Owner be obligated to use more than reasonable diligence in performing the services required of such Owner under this Article VI, be liable for

sidewalks, trons, tree lights, tree rice of a constitution of the control of the

Operator's Platforms upon the Carme and care have a factor and the second of the Exhibit sales and the second of the care and th

Eulldings upon the terms and condition states for the states of the stat

### (4) Inconcionally bullead.

suppression System Suppression Content of the suppression System for the factor content of the factor content

(N) Secondary Condenses Contenses Contenses Contenses Condenses Condense Condenses Condenses Condenses Condenses Condenses Condenses Con

6.2 The Owner of the Assisation regards and concession of the Council and the Council as a second of t

Professional (A) Water bedefiness. Production of which the model of the control o

Mor'dl (as shown on the Flans) and particles of the standard of the flans band and considered of the flans band and considered all apon the terms and considered the frank band considered the frank ban

### 6.3 Intentionally onliced.

Parlities and furwish all make a good-said to realize and furwish all regreece as request to a manner which will provide ear the conjugancy and enjoyment of its request to the intended such as first-classes to testidential property, but in no every early and than responsible diliques to use more than responsible diliques to use more than responsible diliques to use more than responsible diliques to use the required of such owner under this enter that

# UNOFFICIAL COPY 4 3 4

consequential damages for failure to perform hereunder or be liable for interruption or inadequacy of service, loss or damage to property or injury (including death) to any person for any reason. Each Owner obligated to furnish services hereunder reserves the right to curtail or halt the performance of any service hereunder at any reasonable time and for a reasonable period of time to the extent reasonably necessary to perform Maintenance or in an Emergency Situation.

- 6.5 Payment for services rendered pursuant to Article VI hereof and other charges and fees related to such services, including overhead and supervision fees, shall be made in accordance with the terms and provisions of Exhibit 6.5.
- 6.6 (A) If any Owner shall fail to (i) perform the services required to be performed by it pursuant to this Article VI (except when such failure is caused by any other Owner, Unavoidable Delay or such Owner is ertitled to discontinue such service pursuant to Section 6.7 hereof), or (ii) perform its obligations under Section 10.1 hereof, and such failure shall continue for a period of ten (10) days after written notice thereof to the Defaulting Owner from any other Owner, such other Owner shall have the right to perform the same until such time as the Defaulting Owner cures its failure to perform. Such notice shall not be required in an Emergency Situation.
- (B) During any period in which any Creditor Owner is performing pursuant to this Section 5.6, the Defaulting Owner shall make payments to the Creditor Owner as provided in Paragraph 7 of Exhibit 6.5.
- 6.7 If at any time a Defaulting Owner fails to pay to a Creditor Owner any sum of money payable to such Creditor Owner pursuant to the provisions of Section 6.5 hereof within ten (10) days after receipt of written notice from such Creditor Owner demanding payment of said sum of money, then such Creditor Owner may discontinue furnishing to the Defaulting Owner the services for which payment has not been received until said sum of money is paid; provided, however, that if the Defaulting Owner in good faith disputes its obligation to pay said sum of money, pays the undisputed portion of said sum and diligently contests any action or proceeding brought to collect said sum of money or to enforce any lien therefor, or brings an action to determine the respective rights of the parties to such dispute and diligently prosecutes the same, then the Creditor Owner may not discontinue furnishing any such services to the Defaulting Owner unless and until it shall finally be determined by arbitration in accordance with Article XII hereof or by a final non-appealable order of a court of competent jurisdiction that the Defaulting Owner is obligated to pay said sum of money and thereafter said sum of money remains

consequential dimages for failure to server ...

Limite for interruption or inadequacy of ...

to property or injury (includion to the ...

reapon of Each Owner shiptons ...

reserves the right to outled ...

service hereunder at any reasons ...

period of time to the sectors ...

period of time at the sectors ...

Verile 6:2 Payment for services canoered accompay to every hereaff and other kharqes cond seek set of the conincluding creathend and cappary conditions the contraction accordance with the terms and army before a contraction

required to be performed by it pursees when such failure is caused by at pursees when such failure is caused by an except of such openses of the such failure is entitled to discussed in the early and such tailors and failure witten not be attended any caller and any caller one is a such at the same or it is attended to perform the same or it such that it is a such that of perform the same or it such that we same or it such that any caller of perform the same or it such that when the same or it such that it is a such that the same or it such that it is a such that

performing bourdent to the Boules and white and the state of the state

Treeditor Ovrer and any observation of the contraction of the contract pursuant to the provintors of Beecese 1.5 center days, after reseapt of westten notice rese demanding property of weath some to areasy of inibrasis may discortifue formulating to receive the court for which payment has act been been bear near the bold of the **paidi,** provided, chowsveu, sussit to the term of the **faither** to the term of the faither to the term of the faither to the term of the faither the **4**) ' Level Law can bine 30 maistog bedognibny apycitar thecefor, purtrinas an active receptor the result home of kelding sid lo etheric thaysame, then the Cirediter Gymen in a not ... any such agrifican to the Detailed of motiving 11 the contract of the contract o Article XXII hereon or by a limit my a limit of the content IX selected of bompekant just adjustion that the for the last in the first first of to pay soid sum of money and the control of mis bids yed of

unpaid; and further provided, however, that Creditor Owner may not discontinue any such services if such discontinuance would cause an Emergency Situation or hinder steps to remedy or otherwise exacerbate an existing Emergency Situation.

- 6.8 The Owner of the Commercial Building shall have the right, from time to time, to reallocate among the Owners in a fair and reasonable manner the cost of any item of Maintenance under this Article VI if it determines that such cost is not reasonably allocated among the Owners; provided, however, that if any such reallocation has a material impact on any of the Owners, any such Owner shall have the right to submit the issue of the reasonableness of such reallocation to arbitration pursuant to Article XII hereof.
- 6.9 An Cwrer obligated to perform Maintenance of Facilities shall, in replacing Facilities, replace such Facilities with Facilities substantially equivalent or better and providing substantially the same quality of service or better.
- 6.10 If at any time the actual allocation of cost of Maintenance based on an Commer's usage recorded by meters cannot be determined because the meters or system for recording metered information are not installed or operative, then for such period when the usage data from neters is unavailable, the Owner make performing such Maintenance shall make such reasonable determination of costs based on isage, using such experts or performing systems as such Owner may consider helpful to achieve an estimate of usage. Such Owner shall notify the other Owners in detail of its determination of estimated usage and the method for such determination at the time such Owner sends 2 Projection Notice or Statement (as such terms are defined in Exhibit 6.5) or statement of net capitalized cost under Exhibit 6.5 relating to such Maintenance. If, within thirty (30) days after receipt of such notice, the Owners receiving such notice do not in good faith, dispute that such method of estimating usage has been determined reasonably, such determination of usage shall be final and conclusive upon the parties; however, chat said provided, determination shall in any event be subject to the approval of the holders of the First Mortgages; provided further, however, if any Owner receiving such notice, in good faith, disputes that the method of estimating usage has been determined reasonably, he shall so notify the other Owners. If the Owners fail to agree concerning the method of estimating usage within thirty (30) days after receipt of the disputing Owner's notice, then the Owners shall submit the question to the Architect for its determination, which determination shall be final and binding on the Owners, the Owner of the Commercial Parcel and the holders of the First Mortgages.

89208434

unpaid; and Turther provided, however, that the discontinue any such services if such services at such services at such services steps to the discontinue an existing Emorgoody interpretable an existing Emorgoody interpretable.

right, from time to time to real erest and from time to time, to real erest and reasonable manner the control of this Article VI if it determine the cast allocated among the Owners provided the Control has a material large erest and shall have the time to the real location and the real location and the real location that the real location is a fittigle XII horeof.

Single 6.9 An Owner obligated to produce the second of a second analysis of a second analysis of a second analysis of the second and the seco

WWW. 6,10 If Path any ាននៅជា **Bainténance Dosod que do Comerie, o com** And the second to the contract of the contract in tellisteni for was moidamrolni banen the usage data Name and district of the control of leged enter to notifedimentable **Systems** de auch Omnes day 🖒 Buch Owner sons of usage. its determination of estates dateriination at the trae - a te ~ Statument (no such earns are andre a green from Southstines ciencies Maintenanco. J. Withis inich vill nobles, the consistency second and second dispute that such method of achievedor reaconably, anch determination of a conditions when the continues Sandan Branch determination and I in any event led subject nolder of the First Mortgages: provinct the \* Dwnar redeiving buide poticly, in a second of the . The arm and opening prizonides to bodism shall so notify the other canter. . The re**conferming the action of best materials** and in the atternored of the size of the service of the contraction s er er er er er en som og og nolmæskop udde dingue ilinde which determination ghell be find and lesting a true Owner of the Commercial Parcel and the Stewart Morragen.

The Control of the Co

#### ARTICLE VII

#### Compliance With Laws; Removal Of Liens

- The Owners and the Owner of the Commercial Parcel shall each comply with all laws, codes, rules, crders, regulations and requirements now or hereafter enacted promulgated by the United States of America, State of Illinois, City of Chicago and any other governmental or quasi-governmental authority or agency now or hereafter having jurisdiction over the Total Farcel, Building or the any portion thereof, noncompliance would subject the other Owners, the Owner of the Commercial Parcel or any of the holders of the First Mortgages to civil or criminal liability, or would jeopardize the full force or effect of any certificate of occupancy issued to such other Owners or for the Building itself or would jeopardize the Owner of the Commercial Parcel or such other Owner's or Owners' right to occupy or utilize beneficially their respective portion or portions of the Total Property a any part thereof, or would result in the imposition of a lien against any of the property of the other Owners or the Owner of the Commercial Parcel.
- The Owners shall sech comply with all rules, regulations requirements of any insurance rating jurisdiction over the Total Property or any portion thereof and the requirements of any insurance policy affecting insurance coverage on any of the other owners' portion of the Total Property, if noncompliance by it with respect to its portion of the Total Property or any portion thereof would (i) increase the premiums of any policy of insurance maintained by the other Owners or the premiums of any policy of insurance maintained by all Owners, or (ii) render any of the other owners' portion of the Total Property uninsurable, or (iii) create a valid defense to any of the other Owners' right to collect insurance proceeds under policies insuring such other Owner's portion of the Total Property; provided, that this paragraph shall not apply to insurance policies of individual Unit Owners; provided further, however, that if such compliance is hereafter required solely because of the nature of the use, possession or management of or activities in any of the other Owners' portion of the Total Property, such other Owner or Owners shall be liable for the cost and expense of such compliance. If at any time any Owner so obligated to comply shall not proceed diligently with any such compliance and such failure to proceed shall adversely and materially affect any of the other Owners, then any Creditor Owner may give written notice to the Defaulting Owner specifying the respect or respects in which the cure of such noncompliance is not proceeding diligently and, if upon expiration of ten (10) days after the receipt of such notice, any such cure of the noncompliance is still not proceeding diligently, then each

#### ARTICLE VIL

### Compliance With Lawer Reports of Leave

Tell The Owners and the firms of the each scomply with all clave, seekers regulations and requirements now promotories by the United States of a a City at Chicago has bar opioin is vito នាយើងនេះដើម្បា បំណើងជុំមួយប្រាសាលមា មុខ សមាន សមាន សែកកា ស្ត្រី recent that college that the indicating honcompliance would subject the Commercial Pargel or any of the grand off the control of th effect of any certificate of econoor took the sullding transfer or treet Commercial Parcel or rain oth. or a pical velsialization ozilita av the Total Traparty or any park or one generals mail at lor**noisisoqui** Owners or the Owner of the Insurer to

7.2 The owner chail care 1.0 and determinated the fig trya M Juriediction over their tine pequineration 원리견 • • 607 30 governe and approp paksaiquooment di \_ Addresdaged or the prelimen of say pol **Cunera, let (13)** service lang of the second of the secon of their billion of papit, this was noticed and to policies deux "entracam estation 2011 **医原基物位** Buga reache aids dads Property: J. Covided, insurance oplicies of indexidual this each los each trans con air where tower dunk at later a wayewor because who the assure of the use, possed on e. activities, include of the other brach by Eppearty, such other Owner or consocount 3.3 - consilamo, hare to become bas obligated to becapily what anot breeze a sinically erolici dom bas epasilano 化三氯化物 人名英 the contract of the contract of the vilsing and over the lead one of the contract matrices svig van mon it is to be once end dorde at although a priduage a properting diligently and, it soon suplanting aften the secolot of much sonios, buy sour seems es noncompliance is a fill not proceed as consilignment

Creditor Owner may cause such compliance to occur by taking all appropriate steps to carry out the same. Each Creditor Owner shall be entitled to reimbursement upon demand from the Defaulting Owner for all costs and expenses incurred by such Creditor Owner in connection with causing any such compliance to occur.

7.3 Each Owner shall remove, within thirty (30) days after the filing thereof, any mechanics', materialmen's or any other like lien on any other Owner's or the Owner of the Commercial Parcel's portion of the Total Property, or on its portion of the Total Property if the existence or foreclosure of such lien on its portion of the Total Property would adversely affect any Easement created he eunder or services to be furnished pursuant to Article VI hereof, arising by reason of any work or materials ordered or any act taken) suffered or omitted by such Owner. In the event the Defaulting Owner fails to remove any such lien within such thirty (30) day period, each Creditor Owner may take such action as such Creditor Owner may deem necessary to remove such lien. Each Creditor Owner shall be entitled to reimbursement from the Defaulting Owner for all costs and expenses incurred by such Creditor Owner in removing or attempting to remove such lien. Notwithstanding the foregoing, the Defaulting Owner shall not be required to remove such lich so long as within said thirty (30) day period such lien cannot be foreclosed and the Defaulting Owner (A) shall in good faith diligently proceed to contest the same by appropriate proceedings and shall give written notice to each Creditor Owner and the Owner of the Commercial Parcel of its intention to contest the validity or amount of such lien and (B) shall deliver to each Creditor Owner (and the Owner of the Commercial Parcel if the lien is on the Commercial Parcel) either: (i) cash or a surety bond from a responsible surety company acceptable to such Creditor Owner (and to the Owner of the Commercial Parcel and the holder of the First Commercial Mortgage if such Creditor Owner is the Owner of the Commercial Building, to the holder of the First Hotel Mortgage if such Creditor Owner is the Owner of the Hotel Property or to the holde: of the First Residential Mortgage if such Creditor Owner is the Comer of the Residential Property) in an amount equal to one hundred fifty percent (150%) of the lien claim and all interest and penalties then accrued thereon or such greater amount as may reasonably be required to assure payment in full of the amount claimed plus all penalties, interest and costs which may thereafter accrue by reason of such lien claim or (ii) other security reasonably acceptable to such Creditor Owner (and to the Owner of the Commercial Parcel and the holder of the First Commercial Mortgage if such Creditor Owner is the Owner of the Commercial Building, to the holder of the First Hotel Mortgage if such Creditor Owner is the Owner of the Hotel Property or to the holder of the First Residential Mortgage if such Creditor Owner is the Owner of the Residential Property).

Créditor Odner may cance such compliance to occert, a speroprinte star constitue of the star of the st

This commence dissist the control of on the content of the state of the section of the content of the c Course who ha real fails Partiel 's foot to be a second of Islands regarder commenced to the control of rose transport fedot and lo doiding er seriesce to terminate bother or search and applications of Laboratio TV ong becarious cases a dos yas the Belancia territarian prince character yab. (08) grafit e teles control that Goald double has Real Credition Cond. Links that to the Credited Defaulting (Gweds test will Creditor Owner th Franklin Notwith anthrapadity of the madical temper of biriums inger aut a Pale bairsa yan I - D. Hilliam Brook, at 1 I bake (144) appropriate proceedings กราว โดยมา รองพรมี **เมื่อได้เป็นที่เป็** intention to ase two mbha ou revileb lishe Commarcial Parcel in the s sens figure (procure solding design) acteptable to says areation conce Commence Lab Paragraff and the level of the control of it such Creditor paner is the cause of the the nolder of Chartest noted High the last appeared to the the Comer of the Rosel Council of the Experience of Residential Fortgage with such thanking ourse Residential Property in an actual egot on percent (1992) of the lien casis and of the conthan executed thereon or such greates for the contract required to lessouth paymont of the company of the company of jese je en silve sivos bab Tabiotal p**ietificas** reason dat such itien (1) to bis or (1) collection something acceptable to wach (teditor viscor Commercial Parcel and The Relief of the contest for the Tendin Creditor Owner ha také daya sa sa sa ang ang ang ang ang the horder of the Plest meter Mortugge if more f the Council of the Robell Proposity on to the laster of the Barthar Rosella of the Council of th Newldential Property).

Each of the Owners and the Owner of the Commercial Parcel (hereinafter in this Section 7.4, the "Indemnifying Owner") covenants and agrees, at its sole cost and expense, to indemnify and hold harmless the other Owners and the Owner of the Commercial Parcel (hereinafter in this Section 7.4, the "Indemnitees") from and against any and all claims against Indemnitees for losses, liabilities (civil or criminal), damages, judgments, costs and expenses, and any actions or proceedings arising therefrom, by or on behalf of any person, firm, corporation or governmental or quasi-governmental authority, other than the Indemnitees, arising from the Indemnifying Owner's use, possession or management of the Indemnifying Owner's portion of the Total Property or activities therein or arising out of the Indemnifying Owner's use, exercise or enjoyment of an Easement, and from and against all costs, attorneys' ress, expenses and liabilities incurred with respect to any such claim, action or proceeding arising therefrom. any action or proceeding is brought against any of the Indemnitees by reason of any such claim, Indemnifying Owner, upon notice from any such Indemnitees, covenants to resist or defend such action or proceeding with attorneys reasonably satisfactory to Indemnitee. Any counsel for the insurance company providing insurance against such claim, action or proceeding shall be presumed reasonably satisfactory to Indemnitees.

#### ARTICIE VIII

#### Real Estate Taxes

8.1 The Owners and the Owner of the Commercial Parcel agree to file a tax division petition with the Assessor of Cook County, Illinois (the "Assessor") to obtain separate real estate tax parcel identification numbers and separate real estate tax bills for their respective portions of the Total Property, if such a petition has not heretofore been filed. When separate real estate tax bills are received, the Owner of the Commercial Building shall pay the real estate taxes, special assessments and cay and all other taxes and assessments of every kind or nature levied upon the Commercial Property, the Owner of the Hotel Property shall pay the real estate taxes, special assessments and any and all other taxes and assessments of every kind or nature levied upon the Hotel Property and the Owner of the Residential Property shall pay the real estate taxes, special assessments and any and all other taxes and assessments of every kind or nature levied upon the Residential Property. Each Unit Owner shall pay the real estate taxes, special assessments, and any and all other taxes and assessments of every kind and nature levied upon his Unit Ownership.

T. 4 Each of the Consts and the consts to the covenants and agraes, at its cole cost wit spend and hold narmless the other owners and year Partrolo (haireinotteir in this technology of out required the bas yes decleps bas and the contract of the contra explication, and anotical or president on behelf of any person, then, and the quasi-covertmentel duthority, there are rear jobs erroner anilitangobal edijedat Thoesenity ing Gwaer's portion or the c restal of the painting out of the act or enjoyment of the Esseent, ... attorneyst fees, espenses in ite areas to moltos esimio sibue yas Merco of the ballonesson in an adding you by design and are course to displace you any soch Indepoisors vas Aventotis divin publicações **Transc**e Asy constântinos theurance against sure steel problem of reasonably subsection of

The commence was the remain but ince the world better it to fille a tax division petition while the Albert and the Illinois "Lore "measured") to conside Account parcal identification manipule and recorded ich filela gesportiva portional of the rooms and getition has been allocated and basis in the masterial tage tax bills the casestved, the owner of the following pay the real estate taxes, special estates the estates the second of the case of the estates the e the Commedial Property, the Osmur of elected to the the continuence injugate, seaso obstaction add er et la leve de la gaserer do refrencement bins comat The service of the compact educates yaraqual descripthe roal restate teases, epocial convers were the second of the provertion adampeasas billion about Regidential Property | Seen Unit Other and Estimating taxed, special assensagets, but out in it will be a assessments of every kind and namesses Cwaership.

- 8.2 (A) At any time that the Commercial Property, Hotel Property and Residential Property are not separately assessed and taxed, each Owner shall pay its respective portion of such real estate taxes, special assessments and any and all other taxes and assessments of every kind or nature levied upon or with respect to the Total Property. Each Owner's portion of such taxes and assessments shall be determined by the Owner of the Commercial Building in a fair and reasonable manner.
- (B) Upon receipt of the real estate tax bills for the Total Property, the Owner of the Commercial Building shall forward a copy of same to the Owner of the Hotel Property and the Owner of the Residential Property, together with the Owner of the Commercial building's determination of their respective shares of such tax bills. The Owner of the Hotel Property and the Owner of the Residential Property shall each deliver to the Owner of the Commercial Building a cashier's or certified check made payable to the Cook County Collector for its allocable share, as determined by the Owner of the Commercial Building, of the tax bills within five (5) days after demend is made therefor by the Owner of the Commercial Building. The Owner of the Commercial Building shall forward such checks with the Owner of the Commercial Property's share of the tax bills to che Cook County Collector and shall forward a copy of the receipt for same to the Owner of the Hotel Property and the Owner of the Residential Property when it is received.
- obtain a lowering of the assessed valuation upon the Total Property or takes other action for the purpose of reducing taxes thereon with respect to any period prior to the time that the Commercial Property, the Hotel Property and the Residential Property are separately assessed and taxed, the Owner of the Hotel Property and the Owner of the Residential Property shall cooperate with the Owner of the Commercial Property in such attempt and shall each share in the costs incurred in proportion to its share of the real estate taxes. Any tax refund received as a result of such action shall be apportioned between the Owners in accordance with their respective portions of the real estate taxes. Nothing contained herein shall affect the independent right of encl Owner to protest taxes and other charges to the extent the same affect only such Owner's portion of the Total Property.
- 8.3 If, prior to the time separate tax bills are obtained, any Owner shall fail to pay any tax or other charge, or share thereof, which is due and which such Owner is obligated to pay pursuant to this Article VIII, and if such unpaid tax or charge is a lien or encumbrance on the portion of the Total Property owned by any of the other Owners, or if any lawful authority would have the right to sell or otherwise foreclose against the portion of

Froperty and Mandentlal Property are not some the content of the last the content of the content

(B) Upon remain of the Constant of the Copy of same to the Constant of the Copy of same to the Constant of the Residential Englerry.

Voidiercial Building's determined Residential Property of the Commercial Building a captient of the County Collector to the Constant Swinding a captient of the County Collector to the County Collector to the Commercial Epilopic Constant of the Cons

obtain a lowering of L. denomer by the construction of taken of the engage of the enga

any Cwher shall fail to pay by the consequence of the shall fail to pay by the consequence of the shall fail to pay by the consequence of the shall be consequenced to this Archelot VIII, and it was a lient or chemical and the consequence of the consequence of

the Total Property owned by any of the other Owners or extinguish any Easement benefitting any of the other Owners by reason of such nonpayment, or subjects any of the other Owners to personal liability for the same, then each Creditor Owner may, after ten (10) days' written notice to the Defaulting Owner, pay such tax or charge, or share thereof, together with any interest and penalties thereon, and the Defaulting Owner shall, upon demand, reimburse such Creditor Owner for the amount of such payment, including the amount of any interest or penalty amounts accrued thereon.

#### ARTICLE IX

#### Insurance

- 9.1 The Owners shall procure and maintain the following insurance:
  - The Owner of the Residential Property shall (A) keep the Residential Building insured for no less than "all risk" or "special form" coverage on real property and broad form named perils on personal property for an amount not less than one hundred percent (100%) (or such lesser percentage as may be permitted under the Act, but in no event less than ninety percent (50%)) of the "Full Insurable Value" hereinafter defined) thereof. carried by the Owner of the Residential Property hereunder shall also include insurance Residential Easement Ficilities, which insurance shall be primary coverage with respect to any loss to the Residential Easement Facilities, with any insurance carried by the Owner of the portion of the Building in which the Residential Easement Facilities are located being excess coverage, and the Owners shall obtain appropriate endorsements to prevent the insurance companies from raising the claim or defense of co-insurance or other like defenses or like claims adverse to the Owner of the Residential Property.
    - (ii) The Owner of the Commercial Building shall keep the Commercial Building insured for no less than "all risk" or "special form" coverage on real property and broad form named perils on personal property for an amount not less than ninety percent (90%) of the Full Insurable Value thereof (or such greater amount as may be necessary to avoid coinsurance), but, in any event, not less than the greater of the amounts required by the Owner of the

the Total Property owned by any of the other case any Easement Benefitting any of the other case and nonpayment, or subjects and the same, then we will the far the same, then we will days' weither hotier to the test we will days' weither hotier to the test we the test the same the neither to the appear in the same the same to the appear to the same of any interest or panel to any interest or any interest or panel to any interest or panel to any interest or any interest or panel to any interest or panel to any interest or any interest o

#### ARTUGER IR

#### Inaurance

3.1 The Owners shall proper and court

(A) (1) The Couper of the Fact, in the property Residential below the increase the Residential below the property and brown property and brown property and brown percent (1907) and brown percent (1908) and brown the percent (1908) and the percent (1908

(ii) of the Common of the Control of the Control of the Check the Common of the Control of the C

Commercial Parcel and the holder of the First Commercial Mortgage. Such coverage of the Commercial Building shall include coverage for loss of rental income caused by business interruption or extra expense incurred to reduce such loss of income, in such amounts as may be carried by prudent owners of first-class commercial buildings in the City of Chicago, Illinois, or as may be required by the Owner of the Commercial Parcel and the holder of the First Commercial Mortgage. Insurance carried by the Owner of the Commercial Building hereunder shall also include insurance of Commercial Easement Facilities, which insurance shall be primary coverage with respect to any loss Note the Commercial Easement Facilities, with any insurance carried by the Owner of the portion of the Building in which the Commercial Easement Facilities are located being excess coverage, and the owners shall obtain appropriate endorsements to prevent the insurance companies from raising the claim or defense of co-insurance or other like defenses of like claims adverse to the Owner of the Commercial Troperty.

The Owner of the Hotel Property shall keep the Hotel Building insured for no less than "all risk" or "special form" coverage on real property and broad form named parils on personal property for an amount not less than ninety percent (90%) of the insurable replacement cost thereof (or such greater amount as may be necessary to avoid co-insurance), but, in any event, not less than the amount required by the holder of the First Hotel Such coverage of the Hotel Building Mortgage. shall include coverage for loss of income caused by business interruption or extra expense, incurred to reduce such loss of income, in such emonts as may be carried by prudent owners of first class hotel buildings in the City of Chicago, Illinois, or as may be required by the holder of the First Hotel Mortgage. Insurance carried by the Owner of the also Hotel Property hereunder shall include Facilities. insurance of Hotel Easement insurance shall be primary coverage with respect to any loss to the Hotel Easement Facilities, with any insurance carried by the Owner of the portion of the Building in which the Hotel Easement Facilities are located being excess coverage, and the Owners shall obtain appropriate endorsements to prevent the insurance companies from raising the claim or

Commercial Parcel and 一世の社会の計画、協議連 Commercial Serverges. Such Serve Commercial Builting and a task e ga Tempera crasaf **listras la** oerserse sadio Armo Wi The transport the butter 11.15 · 文件设计 666类 作文 that will beginper redifur end 4.37 wit beremse comsument Builted no horse tel. roming of itsea la comande ed od Lylender contained indic stanwolstd edd. ddewolg i to bearings

the social being a constant of the constant of

defense of co-insurance or other like defenses or like claims adverse to the Owner of the Hotel Property.

- (iv) The term "Full Insurable Value" shall mean actual replacement cost (exclusive of cost of excavation, foundations and footings below the lowest basement floor) and shall be determined from time to time by an appraisal prepared by an independent appraiser chosen by the Owner of the Commercial Building, the cost of such appraisal to be shared by the Owners proportionately based on the Full Insurable Value of their respective portions of the Building. Such policies shall be endorsed with a replacement coverage endorsement and an agreed amount clause in accordance with such appraisal;
- (B) Each Owner shall maintain comprehensive general liability insurance wi'n broad form extensions covering claims for personal and bodily injury, death or property damage occurring in, on, within, upon or about (i) the portion of the Total Property owned by such Owner, or as a result of operations thereon (including contractual liability covering obligations created by this Declaration including, but not limited to, those indemnity obligations contained herein), or (ii) any other portion of the Total Property as a result of the actions of such Owner or its lessees, agents or employees. Such insurance shall be primary coverage as to claims for injury or damage resulting from the acts or failure to act of an Owner, with any insurance carried by the other Owners being excess coverage. Such insurance shall be in such amounts as may be required by law and as from time to time shall be carried by prudent owners of first-class commercial, hotel or residential buildings (as the case may be) in the City of Chicago, Illinois, but in all events with limits of not less than \$1,000,000 combined single limit for personal and bounly injury or property damage with an additional \$20,000,000 umbrelia coverage; and
- (C) Each Owner shall insure its boiler and machinery risks, on a comprehensive, blanket basis covering all Building equipment, machinery and apparatus owned by such Owner, consisting of, but not limited to, boilers, heating apparatus, fired and unfired pressure vessels, air conditioning equipment, miscellaneous electrical apparatus and their appurtenant equipment and piping and ducts on a repair or replacement basis for not less than ninety percent (90%) of the Full Insurable Value thereof (or such greater amount as may be necessary to avoid co-insurance), and also providing coverage for loss of income caused by business interruption or extra expense incurred to reduce such loss of

derense of co-insurance or rebor (free discourse). The like claims adverse to the reserve of the

(iv) [ The term limit of a line velue of a line velue of a actual actual replacement of a line of a line of a contact of a line of a line of a line of the contact of a line of the contact of a line of the contact of a line of

to hize throws plant (A) of S ikability insurance with heres for a ober 1. 1. Ber mulich Klibad, base leagedag ons which in a most without the owned by such owner, on (including contractes; lineal) chis Decievation Including, bu obligations contained hereing botali Property as a recult of despect, agentim or emiliary consideration coverege as to claim tell this prince or failure to set of a wine, ligi i min sacran prizag svan**ao vasido** shall be carriet by procest the transfer of the state of the second of t Chicago, this ons, "but he all sware were all the \$1,000,000 combined admic included for concern the property damage with an entitional violation of the

risks, on a comprehensive, hlanist in a comprehensive, hlanist in a comprehensive, hlanist in a comprehensive, hlanist in a comprehensive in a comprehensive in a confined to confined premium premium vessels, of misceliansons electrical appearance and piping and deals on a repair of the principle persion (1904) of the last confinet, persion (1904) of the last confined providing coverage for loss of the also providing coverage for loss of the last confined the expense incurred to the coverage for loss of the last coverage for loss of the cove

income or for loss of use arising from a failure of the Building equipment, machinery and apparatus owned by such Owner or the other Owners, in such amounts as may be carried by prudent owners of first-class commercial, hotel and residential buildings in the City of Chicago, Illinois, or as may be required by the holders of the First Mortgages. Such insurance carried by each Owner shall also include insurance of such Owner's Easement Facilities, which insurance shall be primary coverage with respect to any loss to such Owner's Easement Facilities, with any insurance carried by the Owner of the portion of the Building in which such Easement Facilities are located being excess coverage.

- 9.2 Insurance policies required by Section 9.1 hereof shall be purchased from insurance companies authorized and licensed to transact business in the State of Illinois who shall hold a current Policyholder's Alphabetic and Financial Size Category Rating of not less than A VIII according to Best's Insurance Reports or a substantially equivalent rating from a nationallyrecognized rating service. Insurance policies required by Sections 9.1(A) and 9.1(C) shall (a) be purchased from a single insurance company or group of companies designated by the Owner of the Commercial Building and shall contain the same terms and conditions of coverage and policy wording, and (b) provide for the adjustment of claims with the insurer by the Owner who obtained such policy and, if required by the terms of such Owner's First Mortgage, the holder of such First Mortgage and, with respect to any policy obtained by the Owner of the Commercial Building, the Owner of the Commercial Parcel. So long as the Residential Property remains subject to the provisions of the Act, insurance on additions, alterations, improvements and betterments to individual Units shall be the responsibility of those persons designated in the Condominium Declaration as heing responsible for such insurance.
- 9.3 Each policy described in Section 9.1 hereof: (i) shall provide that the knowledge or acts or omissions of any insured party shall not invalidate the policy as against any other insured party or otherwise adversely affect the rights of any other insured party under such policy; (ii) shall insure as named insured and additional named insureds, as the case may be, the Owner of the Commercial Property, the Owner of the Hotel Property and the Owner of the Residential Property, together with such affiliates of such owners as any of them may designate from time to time, all as their interests may appear; provided, however, that so long as the Residential Property shall remain submitted to the Act, the Association and not the individual Unit Owners of the Residential Property shall be insured as a named insured; (iii) shall provide, except for liability insurance described in Section \*-9.1(B), by endorsement or otherwise, that the insurance shall not be invalidated should any of the insureds under the policy waive

Income or for loss of postation for a state of the equipment, smachinery and apparatus continued for the formatting and apparatus continued for first-class commercial, both as any city of Chicago, Illinois, of the First Montgages, Such in the politicist of the powtion of the Head for the powtion of the Head first the first Montgages are located being occurred.

weet governous half by some mayer Sie will eren en errete romai erromi erroma eta eta eta eta erroma erroma erroma erroma erroma erroma erroma erroma err transact standard in the steel current Follicyholder a chepter ach TON I not less that lo gallas: Reports of a dankingmanlin equality recognized policy service. House (C) I. & Ban (A) I. & caclifood insurance company or group of company the pameralay building and and conditions of coverage and policy adjustment of claims with the Auch policy and, if responds to Mortgage, the holder of accatro entry of bodieson youlday you Owner of the Commercial (\*\* Property resides and each no ademodía . enotribus j no and se Ande other Eschivibat designated in the The Thomas Land U.S. charges a such insuncence.

Racin politicy described in Reaction will n provide that is ned obtained eago or acts of the ner or the real and adapt liver for flate viring of elegant constable has a property by the belief of the constable of the inaural party under sach policy in a contract the 11 Owner of the Conduction is European and Antiand the Owner of the Residenties Program affiliates of such convey as any of but a conto be a fire restable a head less ils fomit for orresported and teach and an paol of a wid the late of the month and board out to a said Residential Property shall be insured as a conservation snall provide, entego tos liabilitos becomes e certies. errole and the result to remember on a semestrobuse Aq 6.142/126 to invalidated should any bill the limiteral control of the land of the control o

MON THE

# UNOFFICIAL COPY 8 9 2 0 0 4 3 4

in writing prior to a loss any or all rights of recovery against any party for loss occurring to the property insured under the policy, if such provisions or endorsements are available and provided that such waiver by the insureds does not invalidate the policy or diminish or impair the insured's ability to collect under the policy, or unreasonably increase the premiums for such policy unless the party to be benefitted by such endorsement or provision pays such increase; (iv) shall provide, except for (a) insurance for loss of rental income or loss of income covered by business interruption or extra expense incurred to reduce such loss of income, and (b) liability insurance required by Section 9.1(B), that all losses payable thereunder shall be paid to the Depositary in accordance with the terms of Article XVII hereof; (v) shall provide for a minimum of thirty (30) days' advance of cancellation, notice non-renewal or written modification thereof to all insureds thereunder and to the holders of the First Nortgages; and (vi) shall include a standard mortgagee endorsement or loss payable clause in favor of each of the holders of the First Mortgages in form satisfactory to it.

9.4 Limits of liability or types of insurance specified in this Article IX or carried by the Owners shall be reviewed by the Owners no less often than annually at least thirty (30) days before the expiration of each policy to determine if such limits, deductible amounts and types of insurance are reasonable and prudent in view of the type, place and amount of risk to be transferred, and to determine whather such limits, deductible amounts and types of insurance comply with the requirements of all applicable statutes, laws, ordinances, codes, rules, regulations or orders and whether on a risk managemen' basis, additional types of insurance or endorsements against special risks should be carried or whether required coverages or encorsements should be deleted. Initially, deductible amounts for insurance required Section 9.1(A), 9.1(B) and 9.1(C) shall not Such limits shall be increased or decreased, \$25,000.00. deductible amounts increased or decreased or type; of insurance shall be modified, if justified, based upon said annual review, and upon any such increase, decrease or modification, the Owners Owner's election, shall, at any execute an instrument increase, form evidencing such decrease recordable modification, which any Owner may record with the Recorder as a supplement to this Declaration; provided that the Owner of the Commercial Building shall have the right, in its reasonable discretion, to increase, from time to time, the limits of liability for the insurance required under Section 9.1(B) and further provided that no agreement regarding a decrease in limits of liability, an increase in the deductible amounts to an amount in excess of \$125,000.00 or elimination of any types of coverages shall be effective without the written consent of the Owner of the Commercial Parcel and the holders of the First Mortgages.

in writing prior to a lower any or all risks of to receive or any party for loss occurring to the conserve energy policy, il such provisions or endered to o erence syd yd neview ddia ddii bebiroug ing the distinct distinct the property of the control of under the policy or management to the log and rebail got in the party of the party of the benefits you long , en la la particiona de la composición del composición de la composición del composición de la composición del composición del composición del composición de la composición r en la leste de la propue de la goldgarredai l**eseuleud** Pust of income, and (b) has general ito send 2. L(B) . that all leasen oursels in the Deposatisary the accordance when the The shall provide for a minimum witten doblee do solden datti w modification thereof to all induces the - ons - trapsydrom - daria codi (to regilo el 2000, exel de decededes **seriestros** the holders of the Piast Forgues a us for

Constituted of the carried by the constitution of the constitution of the carried by the carried Grant of the contract of the c person to miderly and evoled prudent in view of the type that The there were and the first the statement with for a view or engage will be apply bits distances applicable statutes, itself ordinary sleepings organistic de la compania del compania del compania de la compania del compania del compania de la compania del compania d Trisuraring of agelor general e 400 Sept. o an grandad dendager dedakter od die de or receiving a self landente legal alla timber elegate (billing) united that (A) hap in (A) his poblication as bout sas, accorde la such dimina about Control of the deduction of the discressed of decreased to the and a comment of the light made and the best of the contract of the light made and the li and hipon of each riote release, described of the true stall a laterary . Owner's clear that continue . Carryll efdabicoen from evidencing and modification, which any owner may record often supplement to this Declaration; provided the Commercial Childing shall diseas pribilities Islandon discretion; to increase, area cine to iliability from the commence of the comment of the litability, for wise insurence comments of as more and our conservable common and an energy of a specific in a contract it is the contract of the insercass of \$125,000.00 to makingking of the color of the color anull be effective without the weither course is the least Commercial Leardel And the holders of the Ferre Developed .

- 9.5 Copies of all insurance policies, original certificates of insurance evidencing such policies or certified binders delineating all forms of coverage and endorsements required hereunder shall be delivered to each Owner, to the Owner of the Commercial Parcel and to the holders of the First Mortgages at the time of the Conveyances and at least twenty (20) days prior to the expiration date of any such expiring insurance policy.
- 9.6 Should an Owner fail to provide and maintain any policy of insurance required under this Article IX or pay its share of the premiums or other costs for any joint policies, then the other Owners may purchase such policy and the costs thereof (or the Defaulting Owner's share of such costs) shall be due from the Defaulting Owner within ten (10) days after the Creditor Owners' written demand therefor.
- 9.7 Provided that such a waiver does not invalidate the respective policy or policies or diminish or impair the insured's ability to collect order such policy or policies or unreasonably increase the premiums for such policy or policies unless the party to be benefited by such waiver pays such increase, and without limiting any release or waiver or liability or recovery contained elsewhere in this Declaration, each of the Owners and the Owner of the Commercial Parcel, for itself and for each party claiming under, by or through such owner, hereby waives all claims for recovery from the other Owners and the Owner of the Commercial Parcel for any loss or damage to any of its property insured (or required hereunder to be insured) under valid and collectible insurance policies to the extent of any recovery collectible (or which would have been collectible had such insurance required hereunder been obtained) under such insurance policies plus any deductible amounts.

### ARTICLE X

### Maintenance And Repair; Damage To The Building

and 21.7 hereof relating to Maintenance of certain Facilities and areas of the Building or hereinafter in this Article X in the event of fire or other casualty, and without limiting or diminishing each Owner's obligations under Article V, each Owner shall, at its sole cost and expense, keep its respective Property, its Easement Facilities and fixtures, equipment and appurtenances therein (including, without limitation, its portion of the Building facade and its kitchen waste interceptors), in good and safe order and condition, and shall make all repairs or replacements of, in, on, under, within, upon or about such property, whether said repairs or replacements are to the interior

- 7.5 Copies of all thomance colors of current and color instrance oving all forms of colors of colors of colors of all forms of colors of
- of insurance required under the constant of th
- review is doing dudd behivery the respective collicy or policies or distance ability to collect ander and polity or x la de per dens la sanguesta est estada. to be benefited by such walker parlimibing any beleace or weive elsewhere in this Decisoration. the Commorcial Parcel, (Sc under dy or through such ashan recovery from the echer ones Fride Come any loss of damen required hereunder to be totaled assur air of selecting semerusal widenswedle nave bess collection A Company and the Company 2, 5, 71 hereinder/been bath been water week in a war on it is a war deductible emounts

#### X BUDGINA G

### Marchanne and Repairs Danige to the 8 112 cg

and it / hereof as expressly provided to the contract of the Sailding of herein ever the contract of the or other cases to the contract of the or other cases to the contract of the kindred and the contract of the condition, and the contract of the kindred and the contract of the condition, and the contract of the condition, and the contract of the condition of the condition and the condition of the condition.

or exterior thereof, or structural or non-structural components thereof, or involve ordinary or extraordinary repairs or replacements, necessary to keep the same in safe first-class order and condition, howsoever the necessity or desirability thereof may arise, and whether or not necessitated by wear, tear, obsolescense, defects or otherwise, and each Owner further agrees that it shall not suffer or commit, and shall use all reasonable precaution to prevent, waste to such property.

10.2 If the Building is damaged by fire or other casualty and if such damage occurs in, on, under, within, upon or about (a) the Commercial Building only, (b) the Hotel Building only, (c) the Residential Building only or (d) any two such portions of the Building only, and does not in each instance affect the Easement Facilities (1) any other portion of the Building, then any such damage shall be repaired and restored by the Owner or Owners of the portion of the Building in which any such damage occurs in as timely a manner as practicable under the circumstances, and each such Owner shall, in accordance with the provisions of Article XVIII hereof, be entitled to withdraw any insurance proceeds held by the Depositary by reason of any such damage, for application to the cost and expense of the repair and restoration of any such damage. If the nature of the damage is such that it falls within the category set forth in cause (d) above, then the provision set forth in Section 10.3 with respect to the selection of a contractor and the preparation of the plans and specifications shall be applicable. If at any time any Owner so obligated to repair and restore such damage shall not proceed diligently with any repair or restoration of such damage which adversely and materially affects an Easement in favor of any other Owner or services to be furnished any other Owner under Article VI hereof, then (i) each Creditor Owner may give written notice to the Defaulting Owner specifying the respect or respects in which such repair or restoration is not proceeding diligently and, if, upon expiration of ten (10) days after the receipt of such notice, any such repair or restoration work is still not proceeding diligently, then such Creditor Owner may perform such repair and restoration and may take all appropriate steps to carry out the same; or (ii) in an Emergency Situation each Creditor Owner may immediately perform such repair or restoration and may take all appropriate steps to carry out the same. The Creditor Owner in so performing such repair and restoration shall, in accordance with Article XVIII hereof, be entitled to withdraw any insurance proceeds and any other monies held by the Depositary as a result of any such damage, for application to the cost and expense of any such repair or restoration and shall also be entitled to reimbursement upon demand from the Defaulting Owner for all costs and expenses incurred by the Creditor Owner in excess of said insurance proceeds.

or exterior thereof, or structural or rivolve ordinary or replacements, nacessary to keep and condition, however the parties, end whether or rise obsolescense, defect or attraction defect or attraction to series or attraction to preception to prevent or the series or attraction to prevent or the series or attraction to prevent where the restriction to the restricti

Locall al published by S.O. DODO NE if such danage occurs in, un, color 14. Pino pribling informaco Residential Wilding only or Ad the man resem tens, the man selfition Facilities of lany orber poor bening herican od lings openeb the portion of the Entlisher of timely a hegaer on productionic telescoperation, Clarks General Mouse a of bold day od langua ITIVX by the Defortbary by reason the coat and browner of damaga, lak the usatersh the category and the - TO COLUMN TO THE CONTROL idoverprog seld time froducaldege Di Hove vicion ins Tibqei any repair of restorative of materially of sections of Euroscott and envisors of Europe of any of then (1) sach Cooluge Dake Deraulcing Organ rens odu palystse repair or restassion is not a conscinengination of from "(10) days affor Head and the second of the contract such a replace diligently lien soch dradicor eligionation and may remember wittendinan same, or (ii) in an Emergency 1910 immediately perform such repair of aportópiciate stage to decrycest res ser e e performing such repair and rear religion Axticle XVIII hereof, be see to priposeda dund eny bohov sembra com Budishe grany soch danage, for opplication to the coand the board contagnous of the contagnity of the religious ements appear describe thron the Perfection and damped the control of the control control of the insurance proceeds:

10.3 If the Building is damaged by fire or other casualty and if the provisions of Section 10.2 are not applicable because the nature of the damage is such that it does not fall within any of the categories set forth in Section 10.2 then the repair or restoration of such damage shall be the joint responsibility of Said repair and restoration shall be commenced and the Owners. pursued to completion in as timely a manner as practicable and shall be performed on behalf of the Owners by a reputable contractor or contractors experienced in the construction of high-rise structures similar to the Building jointly selected by the Owners. In the event the Owners fail to agree upon the selection of a contractor or contractors, the Owners shall request the advice of the Architect. If after receiving the Architect's advice, the Owners cannot agree on a contractor or contractors, then the selection of a contractor or contractors shall be made by arbitration pursuant to Article XII hereof. The plans and specifications for such repair and restoration shall be prepared by the Architect, unless the Owners otherwise agree, subject to the approval of the Cwner of the Commercial Parcel and the holders of the First Mortgager, in accordance with instructions given by Such plant and specifications shall provide for the the Owners. Building to be rebuilt as nearly as commercially practicable to the Building as constructed prior to the damage unless prohibited by law or unless the Owners and the Owner of the Commercial Parcel otherwise agree, subject to the approval of the holders of the First Mortgages. The Architect shall furnish to each of the Owners, the Owner of the Commercial Parcel and each of the holders of the First Mortgages a set of the plans and specifications which it has prepared or caused to be prepared. Unless the Owners otherwise agree any contractor or contractors shall work under the supervision of the Architect, and the Architect is hereby authorized and directed to instruct the Depositary, from time to time, but only with the prior approval of che Owner in whose portion of the Building such repair and restoration is being performed, and the Owner of the Commercial Parcel and the holder of the First Commercial Mortgage if the Commercial Pailding or any of the Commercial Easement Facilities are involved, the holder of the First Hotel Mortgage if the Hotel Building or any of the Hotel Easement Facilities are involved and the holder of the First Residential Mortgage if the Residential Building or any of the Residential Easement Facilities are involved, as such repair and restoration progresses, to disburse in accordance with Article XVIII hereof, the insurance proceeds held by the Depositary and any other monies deposited with the Depositary pursuant to Section 10.4 hereof for application against the cost and expense of any such repair and restoration.

10.4 If the cost and expense of performing any repair and restoration provided for in Section 10.3 hereof shall exceed the amount of available insurance proceeds, if any, paid by reason of

personal and and one process of bopomed at midling and discusses tt the province of section in a consequence of the section of the nature of the desage is such that it of a section the contegories that fortheir best actions restoration of such Gamage stall be to into content the virgor bisto common soft versely as al actual and of bounding An Monay on bearoung od Albie Geography and desidential and distantings iba oge or reijele aeredoprda exidatyld was not not though out all to the Owners. selection of a formeración es a perm the advice of the Arcidtent. advice, the Owners secure, as the then the selection is a conservation arbived for james up a middeticite egengye goes rod spoldsodiliosga by the Architect, optens the ting approval of the twoner of the land olo altomopopason dan (i okij do the Overs Tack plans and we officer od od pribling Glynasiaeoù en **grifikie**K **edi** comparts exist resolute and well but othexwise agree, subspace First dortgages. reford a see of Owners, the Comerco of the Copy of the First Mortneyes at to Attendance properties of the othervise agree the british supervision of the doed lated ( authorized and diversely to time, but only with the guin portion of the puilding SUNT WARE realist force among the real realist the former as the war of the of the Piron Commercial Mortgage of the trees of the Commondial Basemanc Facilities alo the Piret Rotel Mortgage if the here; warme Edecment Facialities are involved and the Residential Morugage if the Residential in 44 Residential Besement Pacifition and investor restoration progresses, for district the XVIII chereot, the dinaurance procedule have any chief monies deposited with the Branches as monitoring · 🖓 - 1 10,4 hereof for application against the very every such repair and restoration.

restoration provided for in Best Survey of preference on the constraint of the survey of the constraint of the survey of the constraint of

きないではは

## UNOFFICIAL, COPY ...

the damage, then such excess cost and expense (or the entire amount of such cost and expense, if there is no insurance proceeds) shall be borne by the Owners in proportion to the cost and expense of repairing to their former condition their respective portions of the Building; provided, however, that to the extent such excess cost and expense results from the failure of any Owner to maintain the amount of insurance required under Section 9.1 hereof, such Owner shall bear such portion of such excess cost and expense.

19.5 In any instance of repair or restoration pursuant to Sections 10.2, 10.3 or 10.4 hereof, any Owner or the Owner of the Commercial Parcel may require that an estimate of the cost or expense of performing such repair or restoration be made by a reputable independent professional construction cost-estimating firm, except if a construction contract providing for the performance of such repair and restoration for a stipulated sum theretofore has been executed. If said estimate or stipulated sum, or if the accual amount incurred in performing repair or restoration, exceeds the amount of insurance proceeds, if any, paid or payable by reason of the damage (plus any applicable deductible), then any Owner the Owner of the Commercial Parcel or any of the holders of the First Mortgages may at any time give notice to the other Owners demanding that each Owner deposit with the Depositary the amount of such excess cost and expense attributable to each Owner pursuant to this Article X. In lieu of depositing its share of such excess amount based upon said estimate or stipulated sum, or actual cost and expense of performing such repair or restoration, any Owner may deliver to the Depositary security for payment of its share reasonably acceptable to the other Owners, the Owner of the Commercial Parcel, the holders of the First Mortgage, and the Depositary. Such security may be in the form of, but shall not be limited to, an irrevocable and unconditional letter of credit in favor of the Depositary in the face amount of the share owed or an irrevocable loan commitment, satisfactory to the other Owners, the Owner of the Commercial Parcel and the holders of the First Mortgages, issued by a responsible lending institution, to disburse an amount equal to such Owner's share of such excess amount to the Depositary to pay the cost and expense of any such impair or restoration as the work progresses in proportion to such Owner's share of the cost and expense of any such repair or restoration. If the amount of the security required is based on an estimate of the cost and expense of repair and restoration, then the amount of security required to be deposited or available shall be readjusted upward or downward as the work progresses based on actual cost and . expenses of the work. If any Owner shall fail to pay, or, as the case may be, deposit, such Owner's share of the cost and expense (or estimated cost and expense) of performing any repair or restoration in accordance with this Section 10.5, or fails to

the damage, then such excess cost and eppase (secondary country of such cost and expense if there (and expense and expense (and expense of repairing the expense of the Hulling the extent such excess cost and expense to adintain the extent such excess cost and expense (and expense expense of any Owner to adintain the expense (cost and expense) expense (cost and expense)

do some to constant was all sold Sactions 10.2, 10.2 or 19.4 November commenced the focust of the second erregen veggenantentribbag bio laskiasska remittable lindependent or a consequenpel kung sugga – pel kilol dangma (ku**gali**k te to thated doubt ho boneradinish therefore has been eventored aum of delicities actively among restardion, exceeds the paren paid for payable, by presson of deductions, then any owner. any other kalders or the left of notice to the other peners sec the Chepcentering that a secure attributable to canh paner por depositang the chara of the outlinate or ation steed count gerk ve slages hide ghimaoliseg **the** (Depositery security ring **secoptabl**e to the other consul Partiel, ... the distance of the first Such Esquartion may be in the receipt the best andirekvodable ar a unconditioned to establicational Dopositiony in the Trace amount of the court of the loan commitment, satisfactory to the neter the Commercial Farcel and the vollers of in a collegal pailosf aldiagoges, and basel equal to such owners there or our Depositary to pay the rest no current restonation as the work orrections a contract the to appeared bus thou sub to subta and the second of the second with the ୯୯୭ କ୍ରେମ୍ବର ଅନ୍ତର୍ଜ୍ୟ ନିର୍ଦ୍ଧ କେମ୍ବର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ ଅନ୍ତର୍ଜ୍ୟ active to a college deposit perioper ythisuber apparator of operations the sector of the properties elupanses of the mark, the chy tener sinch je enik e rimana) noran (rizakjeti), pri jego prisje (or estimated tost fond expense; of permitted restonation in ancordance with this section

TO STATE OF

# UNOFFICIAL COPY 3 4 3 4

deliver the security provided for herein within ten (10) days after receipt of any other Owner's written demand therefor, then each Creditor Owner may pay the Defaulting Owner's share and the Defaulting Owner shall, upon written demand, reimburse such Creditor Owner for such payment and such Creditor Owner's reasonable costs and expenses incurred in connection with such payment.

damage to the Building, any remaining insurance proceeds paid by reason of such damage shall be refunded to each Owner in proportion to the ratio that the insurance proceeds contributed by reason of such Owner's insurance bears to the total insurance proceeds rade available by the insurer for the repair and restoration, or if the insurance is provided by a single policy covering the building, then the ratio of insurance proceeds attributed to such Owner's portion of the Building by the insurer to the total insurance proceeds made available by the insurer for the repair and resouration. The right of the Owner of the Commercial Building to payment of excess insurance proceeds, if any, shall be subject to the rights of the holder of the First Commercial Mortgage under the First Commercial Mortgage with respect to any such excess insurance proceeds, the right of the Owner of the Hotel Property to payment of excess insurance proceeds, if any, shall be subject to the rights of the holder of the First Hotel Mortgage under the First Hotel Mortgage with respect to any such excess insurance proceeds and the right of the Owner of the Residential Property to payment of excess insurance proceeds, if any, shall be subject to the rights of the holder of the First Residential Mortgage under the First Residential Mortgage with respect to any such excess insurance proceeds, if any, shall be subject to the rights of the holder of the First Residential Mortgage with respect to any such excess insurance proceeds.

10.7 If the Building is destroyed or substantially damaged and the Owners agree not to rebuild, repair or restore the Building, subject to the written approval of the holders of the First Mortgages and the Owner of the Commercial Paicel, then the Building shall be demolished to the extent necessary to comply with all applicable laws, statutes, ordinances, codes, rules, regulations, orders or requirements of any governmental entity or agency thereof having jurisdiction of the Building. In such event, the available insurance proceeds, other than insurance proceeds used to cause said demolition to be performed, shall be refunded to each Owner in the same ratio of insurance proceeds contributed by such Owner's insurance to the total insurance proceeds paid by reason of such damage or, if the insurance is provided by a single policy covering the Building, then in the ratio of insurance proceeds attributed by the insurer to such Owner's portion of the Building to the total insurance proceeds paid by reason of such damage. Such demolition shall be deemed to be a "repair or restoration" to which the provisions of Sections

delivar the security provided for lunery vitain ten en have after receipt of any other owner or vitain receipt of any other Owner's witten as a men any object of a contract owner shall, upon a color of a contract owner whall, upon a color of a color owner which owner was payers and expense a contract owner for any other contract owner of a contract owner.

or in the company of the contract of the contr damaged to be been built to be a considered from the consideration of the consideration of the consideration of no nervo sel finde pphasis della cital accessa engligene ads read along with his holdrogong ringer of general strange down it on charact really and gar of deallaying diving bloop cag in a compressi and it incompaidhroiteas covering the Bullding, cless the rather attracted to seem owner's porterior to the botel insurance proceeds the liepers and restoration. Commercial Building or payror any, estable no autofour of tisks tyns Commercial Morrejage under the respect to any such encase (many) öwder et the notel property ្នាន់ នេះ នេះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ មិនស្រី ខេស្ស ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព្រះ ប្រជាព **denor of the Residential Income to prosec proceeds:** are say, shall its adopent to his enterest of the estimate the contract of the contract of the same than the contract of the contr Mortiguiga With headploon, D. and anch success is not to the con-

and the Owners Lyree not be resulted by the Owners Lyree not be resulted by subject to the vrives approved the first Mortgages and the Crest or the condition and have a statement of a partition of the condition of the condition

10.3, 10.4, 10.5 and 10.8 hereof are applicable except that demolition, and not construction, shall be performed. In the event the Owners agree not to rebuild the Building, subject to the written approval of the holders of the First Mortgages and the Owner of the Commercial Parcel, they may also make provision (i) for sale of the Total Property by the Owners and the Owner of the Commercial Parcel and distribution of sale proceeds, or (ii) for ownership of the Total Property by the Owners and the Owner of the Commercial Parcel as tenants in common, with the right to sue for partition (but for purposes of such partition the Total Property shall be deemed not susceptible of division), all subject to the written approval of the holders of the First Mortgages and the Owner of the Commercial Parcel.

10.8 For purposes of this Article X, architects' and engineers' ress. attorneys' fees, consultants' fees, title insurance premiuns and other similar costs and expenses relating to repair or restoration shall be included in the costs and expenses of any such repair or restoration.

#### ARTICLE XI

### Liens, Debts, Interest and Remedies

11.1 If at any time, any Owner fails within the time period set forth for payment, or if no cime period is set forth, then within ten (10) days after notice or demand to such Owner to pay to any other Owner any sum of money due any other Owner, as Creditor Owner, under or pursuant to the provisions of this Declaration, then, in addition to any other rights or remedies each Creditor Owner may have, such Creditor Owner shall have a lien against the Defaulting Owner's interest in the Total Property and a lien against any insurance proceeds payable to Defaulting Owner to secure the repayment of such sum of money and all interest on such sum accruing pursuant to the provisions of this Article XI. Such liens shall arise immediately upon the recording of a notice by the Creditor Owner with the Recorder and may be enforced by a proceeding in equity to foreclose such lien in like manner as a mortgage of real property in the State of Illinois or by any other remedy available by statute or at law or in equity. Such liens shall continue in full force and effect until such sum of money and any accrued interest thereon shall have been paid in The liens provided for in this Section 11.1(i) shall be subject and subordinate to the lien of any mortgage, trust deed or other encumbrance on the Defaulting Owner's interest in the Total Property at the time of the recording of the notice of lien for all amounts (whenever advanced or accrued) secured by said mortgage, trust deed or other encumbrance, and (ii) are subject to termination and defeat as provided in Section 11.4 below.

demolition, and not construction, addition, and not construction, addition, and not construction, addition approved agree not to require the construction of the Commercial Farce, as the Commercial Farce, and distribute the Commercial Farcel and distribute the Commercial Farcel as tenants in an acceptance of the foregrate as tenants in an acceptance of the foregraphs of the compensation of the Commercial Farcel as tenants in an acceptance of the Commercial Farcel as tenants of the Commercial Farcel as telegraphs of the Commercial Farcel

ingineers (nor purposse of the second of the

Th SUPERINA

articane line stomethic cordet Genelal

and the miner was a past of the control of the cont unt forth for payeons, or if re witchin con (10) days according Carlos of Englishmen amplif within ten (10) days sites to any other over Arediror Cunar, under Declaration, cheo, de esch Cradittor Owner any Breek lien ageinbultos Daf affictor somer and a lieu equipale english english the presentinger and make as an angue ិទ្ធ ប្រទេស ក្រស់នេះប្រទេស នៅអង្គ នយុស្ស interest of when the start result doubt Article You and the contraction of the engineer and the contraction of the enforced by a proceeding in equity to it mannet in the thoughy of the left group of the **by any cottagn domedy, available on acc**ording Such liene shall continue to tull on a con-out of your book your another after a line subject and subgrainabe to the fire or ever talvest gridfunligh with no bonnadapons initiatio Property at the time of the recording of ere, the control of the heart of the control of an action of the Territoria de la competición del competición de la competición de la competición de la competición del competición de la competición de la competición de la competición del competición de la competición del competición del competición del competición del competición del competición del competición d terior and mer those of popioon as issign and police inter-

11.2 To the fullest extent permitted by law, the provisions of Article X of this Declaration shall be controlling over the provisions of the Act insofar as the provisions of the Act purport to limit (i) the obligation of the Unit Owners to repair or restore the Condominium Property or (ii) the use of insurance proceeds for repair and restoration of the Condominium Property. In the event of fire or other casualty or act of God or disaster causing damage to the Condominium Property which would entitle the Owner of the Residential Property, under the Act, to withdraw all or any part of the Condominium Property from the Act and not to repair and restore the Condominium Property as required by this Declaration, notwithstanding the foregoing sentence, then the Owner of the Commercial Building and the Owner of the Hotel Property stall have a lien on the Condominium Property and any insurance proceeds payable for loss or damage to such portion of the Building vacer insurance policies carried pursuant to Article IX hereof, in an amount necessary so that the Owner of the Commercial Building and the Owner of the Hotel Property shall have sufficient proceeds to demolish or repair and restore the Building to a condition so as accountely to assure:

- (a) the structural integrity and safety of the Building;
- (b) the continuous and efficient operation of all Building electrical, utility, mechanical, plumbing and other systems serving the Commercial Building and the Hotel Building;
- (c) compliance with all zoning, building and other laws, rules, orders, ordinances, regulations and requirements of any governmental body or municipality or agency thereof having jurisdiction over the Total Property; and
- (d) the architectural unity and aestrectic appearance of the restored Building as first-class property.

The lien created by this Section 11.2 shall be superior to and take precedence over any mortgage or other encumbrance constituting a lien on the Condominium Property or any portion thereof. Such lien shall arise immediately upon the recording of a notice by the Owner of the Commercial Building or the Owner of the Hotel Property with the Recorder following the occurrence of a fire or other casualty or act of God or disaster stating that it is a lien created by this Section of the Declaration. Such lien shall continue in full force and effect until either the sum of money required hereunder shall have been paid to the Owner of the Commercial Building and the Owner of the Hotel Property, or the Owner of the Residential Property shall have repaired and restored the Condominium Property as required by this Declaration. Such lien may be enforced by a proceeding in equity to foreclose such

III/2/TO the follers extent persitted by the the properties of Article X of this heolementon soull on contraction of provisions of the Act ingerar as the , reserve er en ... to the constant of the continues of the second A 124 1 1 2 response the conducation are property on (1) and the second developed line, gioque **tot abusação** e the so spinesso senso to ealt ip drave ented! causing damage by the Condoninger Economy Const of the Residential Property, this en or lary particler the conductive corpers of the repair and restore the Conductives cargo Declaration, their Plantaning like the con-Ames priblic taloument eur in aman Property shall thave a lice on the even insurance proceeds baved alegatory sonsmusal the building under insurance of its out y and the on amount of the Al Commercial Building was the Osmer of the sufficient proceeds to describe of repolic to a condition so as adequately the accurr

- Buildings, Che Etrusteir at Material Cale Community
- Figure 1 (b) the continuous and article and collection, while the Baildry accusal assistant the Commercial accusations and the Commercial accusations are the Commercial accusations.
- any governmented body or minerally and a constitution of the constitution over the formal body or minerally and the constitution over the formal bady and the constitution over the formal bady or minerally and the constitution over the formal bady or the constitution of the
- Piring (d) (d) the areal tracely and the color of the col

The limit created by the setter and take precedence over an correct take precedence over an correct therefore such lien whall arise impact the foundation of a notice by the Coner of the Concernal the frequency with the Process of the Other Created by this service at a lien created by this service at a lien created by this service at a final force at a service of continue in tall force at a service of Concertal Building and the Concerts the Concerts the Condoninium Property as required by a process of the Condoninium Property as required by a process of the Condoninium Property as required by a process of the Condoninium Property as required by a process of the Condoninium Property as required by a process of the Condoninium Property as required by a process of the Condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property as required by a process of the condoninium Property and the condoninium Prop

lien in like manner as a mortgage of real property in the State of Illinois or by any other remedy available by statute or at law or in equity.

- 11.3 Subject to the limitations set forth in Article XV hereof, and without limiting any equitable remedies to which the Owner of the Commercial Property or the Owner of the Hotel Property may be entitled, so long as the Residential Property remains subject to the provisions of the Act, no Unit Owner shall be personally liable for all or any part of any claim against the Owner of the Residential Property in excess of an amount equal to the amount of the claim multiplied by the percentage of ownership interest in Common Elements allocated to such Unit Owner's Unit Ownership is set forth in the Condominium Declaration. payment of such amount for which a Unit Owner may be liable, (i) any lien arising against such Unit Owner's Unit Ownership on account of such claim shall be deemed released against such Unit Owner's Unit Ownership without further act or deed by any such Unit Owner, and (ii) upon the written request of such Unit Owner, the Creditor Owner who has recorded notice of such lien shall deliver to such Unit Owner an instrument evidencing the release of such lien, but only with respect to said Unit Owner's Unit Ownership. When a Unit ownership is owned by more than one "person" (as defined in the Act; the liability of each such person for any claim against the Unit Ownership shall be joint and several.
- 11.4 No conveyance or other divertiture of title shall in any way affect, diminish or defeat any lien arising pursuant to this Article XI other than a divestiture resulting from a foreclosure of a mortgage lien that is superior to the lien arising pursuant to this Article XI and other than a divestiture of the Owner of the Commercial Building's title to the Commercial Building resulting from the expiration or sooner termination of the Ground Lease, which foreclosure or expiration or termination of the Ground Lease shall automatically cut-off, terminate and defeat any such lien.
- 11.5 The holder of a mortgage or trust deed on all or any portion of the Commercial Property, the Hotel Property or the Residential Property shall have the right to an assignment of any lien affecting the property secured by its mortgage or trust deed, and the Owner of the Commercial Parcel shall have the right to an assignment of any lien affecting the Commercial Building, upon payment of the amount secured by such lien and shall in the event of said payment or satisfaction be subrogated to such other lien and any additional security held by the holder thereof. Such holder of a mortgage or trust deed or the Owner of the Commercial Parcel may at any time give to the holder of the lien a written notice of its election to pay such amount. On a date not less

lion in like mender as a mortuage of reel property of the finite of the colling and in his by sectors of the contract of the security.

The state of the contract of t ner career yes anginet lancative bas Abered Donner to the Commercial Links of the Commercial Property may be entitled, so ins ... The second of the second section of the design of the section of be personally liable for all or ray pour Council to the Rosidential Property in the midding of the clase authorished incorent establish A someth establish dherodal wall in the draph declined addressing e of the for monome date to fading aq tena erese danings spelaton dait. y**jta** géconne es algent elaba de la connection Osnor's Unit Organish william Unit owner (ii) apen the co Survey of the var many, reciber of editions deliyêr tojayeh Katê tween et 🛸 s so drive yino cander, aski ji**qqqë** Ownership. When a best consective consective

11. No conveyance of conveyance of way affect, diminish of Article XI other time of a mortgage lien that a second to the conveyance of the Commercial Colombia Commercial Colombia Commercial Colombia Commercial Colombia Colombia

portion of the holder of a sorty to the Cormerviel of the Cormerviel of the property shall be a soft the Owner of the Cormers of the Amy lies alter of the Amount secured is the third and the correct of the Amount of the Cormer of the Amount of the Cormer of the Cormer of the Cormers of the

than ten (10) and not more than thirty (30) days after such notice of election, the holder of a mortgage or trust deed or the Owner of the Commercial Parcel shall pay the full amount of such lien, and the holder of the lien shall deliver to the holder of a mortgage or trust deed or the Owner of the Commercial Parcel an instrument in recordable form assigning the lien together with the debt secured thereby.

- 11.6 Interest shall accrue on sums owed by a Defaulting Owner to a Creditor Owner and shall be payable from the date any such sum first became due hereunder until paid in full, at a rate per annum equal to the lesser of: (a) the floating rate which is equal to four percent (4%) in excess of the rate of interest from time to time announced by The First National Bank of Chicago at Chicago, Illumois, as its prime rate, reference rate or corporate base rate, or (a) the then maximum lawful rate of interest in Illinois applicable to the capacity of the Defaulting Owner and the nature of the dept. In the event a prime rate, reference rate or corporate base rate is not announced, and no maximum lawful rate applies, then incerest shall accrue at the annual rate of eighteen percent (18%).
- 11.7 Subject to the I mitations set forth in Article XV hereof, the rights and remedies of an Owner provided for in this Article XI or elsewhere in this Declaration are cumulative and not intended to be exclusive of any other remedies to which such Owner may be entitled at law or in equity or by statute. Any Owner may enforce, by a proceeding in equity for mandatory injunction, any other Owner's obligation to execute or record any document which such other Owner is required to execute under or pursuant to this Declaration. The exercise by such Owner of any right or remedy to which it is entitled hereunder shall not precoude or restrict the exercise of any other right or remedy provided bereunder.
- 11.8 Each claim of any Owner arising under this Declaration shall be separate and distinct, and no defense, set-off, offset or counterclaim arising against the enforcement of any live or other claim of any Owner shall thereby be or become a defense, set-off, offset or counterclaim against the enforcement of any other lien or claim.
- 11.9 Actions to enforce any right, claim or lien under this Declaration shall be commenced within three (3) years immediately following the date the cause of action accrued, or such other shorter period as may be provided by law or statute; provided, however, that if prior to expiration of the period in which such action must be commenced, any holder of a First Mortgage is diligently proceeding to foreclose the First Mortgage, then such period in which an action by the Owner of the portion of the Total Property encumbered by such First Mortgage must be commenced shall

than (10) and wot more than thirry (30) may a conformation; the bolder of a moregade of this could be described by the Commercial Parces and the conformation of the the the bolder of the the cold that the conformation of the theodores of the conformation of the course of the course the cold the conformation of the course the cold the cold

Les Interest shall secure or need to seed to seed to seed to some shall received to shall received to shall received to the last of the seed to shall be the last of the seed to seed

11.2 Spifect to the right; and tost of the right; and to the right; and to the right; and tost of the right; and tost of the right; and to the right; and r

or control of the con

Declaration shall be considered until a distriction shall be considered until a decide of the following the content of action are districted as the content of action are districted as may be provided by action and the content of th

be further extended for such additional time as may reasonably be necessary in order for the holder of such First Mortgage to obtain possession of such portion of the Total Property.

11.10 A Defaulting Owner shall pay the reasonable attorneys' fees and court costs paid or incurred by a Creditor Owner in successfully enforcing its rights against the Defaulting Owner under this Declaration.

#### ARTICLE XII

#### Arbitration

- 12.1 The following matters shall be submitted for arbitration to the American Arbitration Association (the "AAA") pursuant and subject to the provisions of this Article XII:
- (a) All disputes, claims or controversies arising under this Declaration involving an amount not exceeding \$200,000 (in 1988 equivalent dollars) which shall not be resolved within sixty (60) days after same have arise; and
- (b) All other matters which are required under the provisions of this Declaration to be submitted for, or determined by, arbitration.

Any such dispute, claim, controversy or matter is referred to herein as a "Matter". Arbitration of any Matter shall be initiated by any Owner making a written demand therefor by giving written notice thereof to the other Owners and by filing a copy of such demand with the AAA. The AAA shall have jurisdiction upon the giving of such notice and the filing of such demand. Any such arbitration shall be held in Chicago, Illinois, and shall be conducted and completed in an expeditious manner and without delay. The Owner of the Commercial Parcel shall be a party to any arbitration of a Matter involving a matter which requires the consent or approval of the Owner of the Commercial Parcel hereunder. The holders of the First Mortgages shall be party to any arbitration of a Matter involving a matter which requires the consent or approval of the holders of the First Mortgages hereunder.

12.2 Unless otherwise agreed to in writing by the parties to the arbitration, within twenty (20) business days after the notice demanding arbitration has been given, the parties shall jointly designate one arbitrator to resolve the Matter. If the parties fail to designate the arbitrator within such time period, an arbitrator shall be appointed in accordance with the procedures set forth in the applicable AAA rules, provided, however, that in

be further extended for such additional and such additional and such for the holder of the such portion of

11710 A Palaulting Owner stall relations of the court can court court court this policy regard against this Declaration.

#### TIN STOTTAM

#### antigord (daA

to the American Arbitration Area at all as such as Subject to the provintence of the Area at a second of the provintence of the Area at a second of the provintence of the Area at a second of the provintence of the Area at a second of the Area at

provision of this perlanguage as a second of the provision of this perlanguage as a second of the perlanguage as a second of

Any such dispute, claim concern berein as a "Matter" ker continued by any come as no rectantiation between with the continued by continued by the continued of section and relief the conducted and realist the conducted and realist the conducted and realist the conducted and realist the continued by the continued of the continued of the consent of approval of the bereander. The holders of the section of a Matter law is any arbitration of a Matter law; is a sectional of the section of a Matter law; is a section of the section of a Matter law; is a seconomic law; is a section of a Matter law; is a section of a Matt

the arbitration, within twenty demanding arbitration has been added arbitration has been accommon to the continuate one arbitrator to the continuation has arbitrator to the arbitrator shall be appointed to the continuation and the applicable AAA research.

any event such arbitrator shall be experienced as to the design, construction and/or operation, as the Matter requires, of high-rise, multi-use structures similar to the Building. Except where contrary to the provisions set forth in this Declaration, the rules of the AAA for commercial arbitration shall apply to the arbitration of any Matter. During the twenty (20) day time period referenced above, the parties may agree in writing to any additions, deletions or changes to the applicable arbitration rules.

- 12.3 The arbitrator shall commence hearings within sixty (60) days of selection, unless the Owners or the arbitrator agree upon an expediced or delayed schedule of hearings. Prior to the hearings any Owner may send out requests to compel document production ron the other Owners. Disputes concerning the scope of document projuction and enforcement of the document requests shall be subject to agreement by such Owners or may be ordered by the arbitrators to the extent reasonable. The arbitrator may obtain independent legal counsel or other professional consultants to aid in resolution of legal or other questions presented in the course of arbitration to the extent reasonably necessary to the fair resolution of the Matter and to the extent that it is economical to do so considering the financial consequences of the The arbitrator in rendering a decision may base such Matter. decision only on the facts presented in the course of arbitration and shall not modify or amend the provisions of this Declaration. Subject to the other terms hercof, if any Owner (or, where applicable, the Owner of the Commercial Parcel) fails or refuses to appear at and participate in an arbitration hearing after due notice, the arbitration panel may hear and determine the Matter upon evidence produced by the appearing Owners. The arbitration costs shall be borne equally by each Owner (and, where applicable, by the Owner of the Commercial Parcel), except that each Owner (and the Owner of the Commercial Parcel) shall be responsible for its own expenses.
- 12.4 Unless otherwise agreed in writing, the Comers shall continue to perform all obligations and make all payments due under this Declaration in accordance with this Declaration during the course of any arbitration constituted or conducted under the provisions of this Article XII. The obligation of the Owners to continue performance and make payments despite the existence of an arbitration hereunder shall be enforceable by any party to the Matter by application to any court of competent jurisdiction for an injunctive order requiring the immediate performance of such obligations as provided in the preceding sentence until such time as any matter is resolved as provided in this Article XII.

any eventureuch arbitrator shall he apperlenced as the constitution and/or operation east the desire and the provisions of the fall and the fall and the contrary to the provisions as to the test and the provisions of the appearance as the contrary of the AAA for security to the contrary of the AAA for security to the contrary arbitration of any fatter. During the there we referenced above, the parties are appearanced above, the parties are appearanced above, the parties are appearanced above.

The remove of the second of th days of gelection, anics the unority of the an expedited or idelayed activities of hearth of hearth and any summer way beard out the production from the exter Overer, when eservation has automborg dasmapobats compared by stopped to agreement by compared light the arbitrators to the entrol case. obtain leaneles legal libraradah dipitos to aid in resolution of legal of other contra ant at holdersides to carrot tair resolution of the watter acchemical to be so considering the ar all more sublices a factor decision properties the properties of special term and present to vision for Italia bas Company and the second Subject the other tends hereof to the and selection of the contraction or the file of the relative the minding privile case the tranger. Or nothing and carbitritizing proof our deal of the deal of the apon evidence produced by the appearant concess, are the concess. coate shall the Borre galatily by oach dwar coat warre and a sour dy ithe constant of the compainted is the real compainted to the constant of the constant is a constant of the (and the Owner of the gonnercial tencest) and the error of the its own expendes. The

continue to perform ail obliquings according to the continue to perform ail obliquings and continue this beclaration in nucerias a very content of any arbitration of any arbitration of any arbitration of this article MET. The continue performance and make payanne description derivation to the content of t

12.5 With respect to any Matter subject to arbitration under this Article XII, it is agreed that the arbitration provisions of this Article XII shall be the sole remedy of the Owners (and the Owner of the Commercial Parcel) under this Declaration. Notwithstanding any other provisions of this Declaration, the foregoing agreement to arbitrate shall be specifically enforceable under prevailing arbitration law. The foregoing agreement to arbitrate shall not constitute any agreement or consent to arbitration of any dispute, claim, controversy or matter not described in this Article XII or with any person not named or described herein, provided that any arbitration proceeding initiated under the terms of this Article XII may, at the request of any party, be joined or consolidated with other arbitration proceedings involving additional parties if the Matter and the subject of such other proceedings arise out of common or interrelated ractual occurrences. Any award of the arbitrators shall be final and binding upon the Owners, the Owner of the Commercial Parcel and the holders of the First Mortgages and judgment thereon shall be entered by any court having jurisdiction.

12.6 For purposes of this Article XII, "1988 equivalent dollars" means the equivalent purchasing power at any time of the value of One Dollar (\$1.00) in calendar year 1988. The 1988 equivalent dollars of any amount shall be determined by multiplying said amount by one (1) plus a fraction, the numerator of which is the difference betweer (x) the monthly Consumer Price Index (as hereinafter defined) last published prior to the date of such determination and (y) the Consumer Price Index for December, 1988, and the denominator of which is the Consumer Price Index for December, 1988. As used herein, the term "Consumer Price Index" shall mean the Consumer Price Index for Usan Wage Earners and Clerical Workers, City of Chicago, All Items (Base Year 1967 = 100) for the applicable month published by the Bureau of Labor Statistics of the United States Department of Tabor or similar index agreed to by the Owners if such index is no longer available.

#### ARTICLE XIII

#### Unavoidable Delays

The Owners shall diligently perform their respective obligations set forth herein. No Owner shall be deemed to be in default in the performance of any obligation created under or pursuant to this Declaration, other than an obligation requiring the payment of a sum of money, if and so long as non-performance of such obligation shall be directly caused by fire or other

Taisonith prepart to any Marter subject on a live of rain Artiste XII. di is agreed that the coloridad mini this Article XII shall be the cole remed of Like IX slotted with Ownerd Office the Commercial Charces) and the country Wolwithstanding any other provides of the foregoing agreement to aithitrate thank under prevailing arbitration but van briditanco bon liene startidus arbitration of any dispute, claim, continue described to this Article All or although the continue of the cont described berein, provided tone 114.637 file a mind in leasest par isone bedelthal of any party, ben joined or concept the livers proceedings throughby addresses gentlessorg compagned which a doubt los footba 5.5 A course this mous paids id boss land 2 as lists (Committee) Parteel and the business of nosuoni Inerech Designing 1 od Ilnia jurisdiction.

12.6 for purposes of this A. ...

Rollers means the equivalent uses no gaivelet follers of any and this extract follers of any articles is the fifteeness because of the second of the s

### ATTER MIDITIA

### Lystal wideblovenu.

The comers shall disgest; seed to the control obligations set forth herein. We control obligations set to the performance of set obligant to this beclaration, other when the payment of a seed of and is the payment of a seed of sective of sective of such collection shall be directly of section of such collections.

casualty, national emergency, governmental or municipal laws or restrictions, enemy action, civil commotion, strikes, lockouts, inability to obtain labor or materials, war or national defense pre-emptions, acts of God, energy shortages or similar causes beyond the reasonable control of such Owner ("Unavoidable Delay") and the time limit for such performance shall be extended for a period equal to the period of any such Unavoidable Delay; provided, however, that the Owner unable to perform (the "Non-Performing Owner") shall notify the other Owners in writing of the existence and nature of any Unavoidable Delay within ten (10) days after the onset of any such Unavoidable Delay. The Non-Performing Owner shall, from time to time upon written request of any of the other Owners or the Owner of the Commercial Parcel, keep the other Owners and the Owner of the Commercial Parcel fully informed, in writing, or all further developments concerning any such Unavoidable Delay and its non-performance.

## ARTICLE XIV Condemnation

- 14.1 In the event of a taking by the exercise of the power of eminent domain or deed in lieu of condemnation of all or any part of the Total Property by any competent authority for any public or quasi-public use, the award, damages or just compensation (the "Award") resulting from any such taking shall be allocated and disbursed, and any repair and restoration of the Building shall be performed, in accordance with the requirements of this Article XIV.
- 14.2 All Awards resulting from the taking of all or any part of the Total Property, other than damages resulting from a taking of the temporary use of space as hereinafter described, shall be paid to the Depositary and disbursed by the Depositary as hereinafter provided. In the event of a taking or a comporary use of any space not including Residential, Hotel or Commercial Easement Facilities or affecting services described in Sections 6.1, 6.2 or 6.3 hereof, each of the Owner of the Commercial Property, the Owner of the Hotel Property and the Owner of the Residential Property shall be entitled to receive directly from the taking authority any Award resulting from such temporary taking within its respective portion of the Total Property according to the law then applicable.
- 14.3 In the event of a taking (other than a temporary taking) of a part of the Commercial Property only, the Hotel Property only, the Residential Property only or any two such portions of the Total Property only (not including any Easement Facilities of any other portion of the Total Property or affecting services

casualty; aational conorgefeet, economism of the constitution restrictions; enemy section, civit commens, restron Line for the second labor of redefinition of the second state of the second second second second second second promotions, acts of God, energy theretally acts beyond the Ressonable common of some ways the season and the time limit for such percent on the period aqual to the period or provided, bowever, that the cases are Performing Owners) while modify the control existence and nature of any Ondersidable never Delining and State of the State after the onest of any such branchation of ្រុកការ ( a.ស. 1900) ស្រាក់ស្រាក់ បានប្រជាពល់ មាន នៅក្រុម ប្រើប្រាស់ ប្រាស់ ប្រែសាធារី ប្រែសាធារី ប្រាស់ ប្រាស cther Owners of the Owner of the Council were a little comport ests to gother and this examed wolfant, in at all there Unavoidable Delay and its movement of acc

of the take event of a tehing eminent demain or deed in lieu f lander of the fire by the the Tobes will be an account to the Tobes will be a considered to the top of the Tobes will be a considered to the Tobes will be a considered to the Tobes will be a considered to the top of the Tobes will be a considered to the top of the Tobes will be a consid quasi-public one. the award, or good at the "Avard") resulting from the condition of this continuent istri in chestitoper sed dile conservationi c**hestiticq** 

**if the Notice Property, chain also demonds** of a the tompostry use of eggeod in him to the gladin - for street is the five in the first condition of the first 14 nerge as the conservation of the story of the start and services and continues ( political dom political Marks ្<sup>រិប</sup> The second of the second transfer of contilinos Januares o grande de la compacta del compacta del compacta de la compacta del la compacta de la compacta Property and Owner of the Horel Property of the english Residential Property shall be covered to we save the "tabling jauthority jeny" Averu "resultion there were come taking julihin oles dangentive paraelang ada da aba abin mini 1. according to the law then application

ng kina nga kanggalah na ali makiya ngahiya ngapiyang na **Tib. Tinanga mali**ng**ita (1.744** of a part of the Commercial English will a selfunity ) the Residentifet timecaty out to account the definition of the Cotar property willy (not since was may meet a survive or a survive was the constant the constant that the constant or the constant that the constant or t